5 Bulgarian

Ernest A. Scatton

1 Introduction

Bulgarian is the national language of the Republic of Bulgaria – the native language of its ethnic Slavonic majority. The estimated population of Bulgaria in 1986 was close to 9 million, nearly 85 per cent of whom were recorded as ethnic Bulgarians. Modern Bulgarian directly continues the Slavonic dialects spoken in the eastern Balkan Peninsula from the time of the arrival of Slavs in the middle of the first millennium AD. It is first recorded in the earliest Slavonic, that is Old Church Slavonic (Old Bulgarian), manuscripts.

Four periods are customarily distinguished in the history of Bulgarian: (1) the **prehistoric period** (essentially Proto-Slavonic), from the time of the Slavonic invasion of the eastern Balkans to the Cyrillo-Methodian mission to Moravia in the ninth century; (2) **Old Bulgarian** – the ninth to the eleventh centuries, reflected in Old Church Slavonic manuscripts; (3) **Middle Bulgarian** – from the end of the eleventh to the beginning of the fifteenth century, a time of rich literary activity and major structural innovation; (4) **Modern Bulgarian** – from the end of Middle Bulgarian to the present, including the years of Ottoman domination (early fifteenth century to 1878).

While the early stages of a number of major innovations are perhaps attested in Old Church Slavonic texts, the most significant changes in the evolution of Modern Bulgarian appear to have begun in the Middle and early Modern periods. Details of these changes are not reflected systematically in written records. The penetration of vernacular features into the written language was impeded for a number of reasons, most importantly conservative scribal attitudes and various **orthographic** reforms which artificially normalized scribal practices during Middle Bulgarian.

Written records from the first two centuries of Modern Bulgarian are quite limited owing to the Ottoman conquest, which severely curtailed the rich literary activity of earlier years. After approximately 1600 – the time of the beginning of the Bulgarian Възраждане/Văzraždane 'Renaissance' – increasingly numerous vernacular intrusions into popular texts suggest that the major changes differentiating Modern Bulgarian from its pre-

decessor were essentially complete. In so far as continuing conservative orthographic conventions make textual evidence problematical, we can only reconstruct the general lines of the actual processes of these changes.

The form of the modern Bulgarian literary language began to take shape only towards the middle of the nineteenth century. Earlier, within the prolific literary activity that developed from the beginning of the nineteenth century, three different orthographic positions competed with one another: (1) conservative, advocating a modern literary language based on the Russian/Serbian recension of Church Slavonic used in the Bulgarian Orthodox Church at that time; (2) progressive, calling for a literary language based on the speech of one or the other of the most influential regions of nineteenth-century Bulgaria; (3) intermediate, advocating use of those features shared by all regional dialects plus Church Slavonic features whenever the dialects differed. The emergence of north-eastern Bulgaria in the middle of the nineteenth century as the nation's cultural, economic and political centre led to wide acceptance of north-eastern dialects as the basis for the literary language. Of these dialects, that of Tărnovo, one of north-eastern Bulgaria's most important cities and the country's first modern capital, was taken as the model for the emerging literary language. On 5 February 1899, the Bulgarian Ministry of Education officially codified the modern literary language with the adoption of the first orthographic system sanctioned for the entire nation. Reformed to any significant extent only once, in February 1945, this system remains in use today.

Despite the fact that the modern Bulgarian literary language reflects the north-eastern dialect of Tărnovo, it incorporates a number of non-eastern, western features (the major dialect division of Bulgarian is east versus west; see section 6). Originally, 'westernisms' were due to adoption of a number of Church Slavonic features coincidentally identical to features of western Bulgarian. However, with the shift of the Bulgarian capital from Tărnovo to Sofia (in the centre of western Bulgaria) in 1879 and the latter's rapid growth into the nation's single most influential urban centre, literary Bulgarian began a gradual process of accommodation to the native dialect of the new capital. As a result, the official literary language – described in standard reference works (Георгиева, Станков/Georgieva, Stankov 1983; Тилков, Бояджиев/Tilkov, Bojadžiev 1981; Граматика/ Gramatika 1982–3; Пашов, Първев/Раšov, Părvev 1979; Романски/ Romanski 1955–9; Чолакова/Čolakova 1977–90) – represents no naturally occurring regional dialect. Rather, it is an artificial hybrid, learned to a greater or lesser extent by all Bulgarians in the course of their schooling.

Prescriptively correct Bulgarian is **spoken** only under the most formal circumstances and only by speakers particularly concerned with proper usage. In less formal circumstances, non-literary features characteristic of speakers' regional dialects frequently occur. The speech of many educated

Bulgarians represents a continuum, with the colloquial, non-literary speech of their native regions at one end and the learned, literary standard at the other. In actual usage, speakers move back and forth between these two poles, incorporating, to various degrees, non-literary features into their formal speech and vice versa. Since the Second World War, owing to the rapid growth of the population and cultural prestige of Sofia, the westernized conversational speech of educated natives of the city has gained increasing prestige and has come to be regarded by some linguists as a (if not the) standard spoken variant of the literary language.

2 Phonology

2.1 Segmental phoneme inventory

Table 5.1 Vowel phonemes

	Front	Central	Back	
High Mid Low	i e	ă a	u o	

/i e u o/ are relatively lax. / \check{a} / is tenser and higher than [a], but lower than Russian [\check{i}]. Word-initial vowels are preceded by a glottal stop.

Unstressed vowels are shorter and weaker than their stressed counterparts, and approach one another pair-wise, /i ·> e ă ·> a u ·> o/, without merging completely – at least not in careful literary speech (Γραματικα/Gramatika 1: 132). Unstressed word-final vowels are often voiceless.

Vowels are **nasalized** before nasal consonants followed by fricatives; often the nasal consonant is lost: о́нзи/önzi [önzi] ~ [özi] 'that (м sG)'. In native Bulgarian words vowel sequences occur only across morpheme

In native Bulgarian words vowel sequences occur only across morpheme boundaries: знаеш/znaeš /zna-e-š/ 'know-PRS-2 SG'. In borrowings they are common: театър/teatar 'theatre'.

/t d s z c l/ are alveo-dental, /n r/ alveolar. The acute accent (') marks palatalization in labials and alveolars. Alveo-palatal obstruents are weakly palatalized. Palatalized labials are pronounced by many Bulgarians as sequences of [Cj]; palatalized alveo-dentals, especially /n' l' r'/, may shift towards alveo-palatals.

/n/ has a velar allophone, [ŋ], before velars: гатанка/gatanka [gatanka] 'riddle'.

/l/ is markedly velarized except before front vowels.

/c/ and /x/ have voiced allophones [3] and [y] before voiced obstru-

	Bilabial	Labio- dental	Alveo- dental	Alveo- palatal	Palatal	Velar
Stops	р р′		t t'		k′	k
-	p p' b b'		d d′		gʻ	g
Fricatives		f f'	s s'	š	•	X
		v v′	z z'	ž		
Affricates			c c'	č		
				ž		
Nasals	m m′		n n'	Ü		
Laterals			1 1'			

Table 5.2 Consonant phonemes

Trills

Glide

ents (except $/v \, v'$; see below). /x/ has a palatal allophone [x'] before front vowels (see below).

j

r r'

Some inventories of Bulgarian phonemes include /x' 3 3'/. However, /x' 3'/ occur only in foreign proper nouns, for example, Xюм/Хји́ш /x'úш/ from English Hume, Ядзя/Jadzja /ja'a/ from Polish Jadzia. Besides foreign proper nouns, /3/ occurs in a small number of non-literary, dialectal words which, if used in the literary language, regularly replace /3/ by /z/: дзифт/dzift ~ зифт/zift 'tar'.

Obstruents contrast voice before sonorants and /v v'/: δόδ/bòb /bòp/'bean' versus πόπ/pòp /pòp/ 'priest', τβόμ/tvòj /tvòj/ 'your-SG (M SG)' versus μβόρ/dvòr /dvòr/ 'yard'. Otherwise they are voiceless word-finally or before voiceless obstruents, and voiced before voiced obstruents.

Palatalized and non-palatalized consonants contrast only before non-front vowels. Moreover, in native words they contrast before /ā u o/ only across morpheme boundaries: xòμπ/xòdja /xòd'-ā/ '(I) walk' (= 'walk-1 SG') versus xòμα/xòda /xòd-ā/ 'the course' (= 'course-the'), бяπ/bjal/b'al/ 'white (M SG)' versus баπ/bal/bal/ 'ball (= dance)'. Otherwise – word-finally, before consonants or before front vowels – they are non-palatalized.

Palatals and velars contrast before non-front vowels; palatals occur here only in words of foreign origin: ron/gol/gol/'naked (MSG)' versus Γροπ/gjol/g'ol/'puddle' (from Turkish). Otherwise palatals are regular before front vowels, velars elsewhere.

After vowels /j/ occurs at the end of words or before consonants (мой/moj 'my (м sg)', дайте/dajte '(you-PL) give!'); before non-front vowels it occurs word-initially or after vowels (язва/jazva 'ulcer', моя/moja 'my (F sg)'). It neither follows consonants, nor precedes front vowels – except word-initially in a few borrowings (йезуйт/jezuit 'Jesuit').

Geminate consonants occur only across morpheme boundaries: отдел/

otdel /od-del/ 'department, section' (= 'of-part'). Consonant clusters are subject to other constraints (Граматика/Gramatika 1: 135ff.).

The letters of the Bulgarian alphabet and their standard transliteration according to the Bulgarian Academy of Sciences are shown in table 5.3.

Table 5.3 Bulgarian alphabet

а	a	K	k	ф	f
б	b	Л	1	x	h [here transliterated x]
В	v	M	m	Ц	c ,
Γ	g	Н	n	ч	č
Д	d	0	o	Ш	Š
e	e	П	p	Щ	št
ж	ž	p	r	ъ	â [here transliterated ă]
3	Z	С	S	Ь	j [here transliterated '; occurs only in ьо/'o]
И	i	T	t	ю	ju
Й	j	у	u	Я	ja

Except for the details that follow, orthographic spellings give a close approximation of phonemic representations (tables 5.1 and 5.2). However:

- 1 щ denotes /št/. / $\frac{3}{3}$ / is written дж/dž. To the extent that it occurs at all, / $\frac{3}{3}$ / is written дз/dz.
- 2 Spellings show the **underlying** (morphophonemic) voicing of obstruents; compare 'alive' M SG жив/živ /žif/ versus F SG жива/živa /živa/.
- 3 Palatal and palatalized consonants before non-front vowels are indicated by я (for /a/ and /a/), ю (/u/), ьо (/o/). Two of these symbols, я and ю, also indicate /j/ followed by /a a/ and /u/, respectively (see examples above). Otherwise, /j/ is written й: мой/ moj 'my (м SG)'.
- 4 The letter я is ambiguous. Most frequently it represents /a/ after palatals, palatalized consonants or /j/. It is also used in two morphological categories for /ă/ after palatalized consonants or /j/: (a) masculine singular definite article, for example, денят/denjat /den'āt/ 'the day (SBJ)'; (b) first person singular/third person plural present tense forms of first-/second-conjugation verbs: xòдя/xòdja/xòd'ā/ 'I walk', броят/brojāt/brojāt/ 'they count'.
- 5 Similarly, a is ambiguous; in two categories it marks /ă/, not /a/: (a) objective (not subjective) masculine singular definite article of nouns: града/grada /grada/ 'the city (M SG OBJ)'; (b) first person singular/ third person plural present forms of first-/second-conjugation verbs: пекат/рекат /pekat /pekat/ 'they bake'.

/ā/ is the regular Bulgarian reflex of PSl. *p and 'strong' *ъ: *pọtь път/pāt 'road', *zъlъ зъл/zāl 'evil (м sG)'. Later in the history of Bulgarian, /ā/ was inserted to break up stem-final consonant clusters terminating in liquids or nasals ('epenthetic ă'): *dobrъ добър/dobār 'good (м sG)' (compare F sG добра/dobra). /e/ is the regular reflex of PSl. *p and strong *ь: *pętь пёт/рёt 'five', *dьпь дён/dёn 'day'. Weak *ъ/ь were generally lost: *zъla зла/zlà 'evil (F sG)', *dьпі дни/dni 'days'. However, weak *ъ/ь remained in a number of monosyllabic roots (Velcheva 1988: 146-8): дъно/dāno 'bottom' from *dъпо, пёсове/рёsove 'dogs' plural of пёс/рёs from *pьsъ. There is a small number of examples in which PSl. *ь, weak or strong, gives /ā/ ('Umlaut of jers', Velcheva 1988: 136-9): *pьпь gives пън/рап 'stump', *tьта gives тъма/tăma 'darkness'.

Bulgarian has lengthened and metathesized reflexes for Proto-Slavonic liquid diphthongs with *o/e:

*gordъ град/grad 'city'
*bolto блато/blato 'swamp'
*bergъ бря́г/brjag 'shore'
*melko мля́ко/mljako 'milk'

Proto-Slavonic liquid diphthongs with $*_{\mathcal{D}/\mathcal{D}} + l/r$ merged with sequences of $*_{l/r} + {_{\mathcal{D}/\mathcal{D}}}$ with loss of the distinction between ${_{\mathcal{D}}}$ and ${_{\mathcal{D}}}$. Subsequently $/_{\tilde{a}}/$ developed in all instances, before or after the liquid depending on the following consonantal environment: liquid-vowel before two consonants, vowel-liquid before consonant-vowel. This is the origin of the **metathesis** alternation described in section 2.3 (see examples there).

PSl. *tj (and *kt + front vowel), *dj gave /st žd/:

*světjь свещ/svěšt 'candle' *noktь нощ/nošt 'night'

*medja межда/mežda 'boundary'

Bulgarian word stress is dynamic: stressed syllables are louder and longer and have a higher fundamental frequency than unstressed syllables. Stress is free: it may fall on any syllable of a polysyllabic word. It is mobile: its position may vary in inflection and derivation. Compare:

'city' sg град/grad : sg DEF градът/gradat : PL градове/gradove 'bridge' sg мост/most : sg DEF мостът/mostat : PL мостове/mostove

Stress is distinctive: вълна/vålna 'wool' versus вълна/vålna 'wave'. For stress patterns of major lexical categories see section 3.

Simple Bulgarian words have a single stressed syllable. Compound words may carry a secondary stress, or even two stresses of equal prominence; for example, най-добър/naj-dobar 'best' (= 'most-good'),

силнотоков/sìlnotokov 'high-tension (M SG ADJ)'.

Bulgarian distinguishes three small classes of inherently stressless clitics: proclitics, enclitics and variable clitics (either enclitic or proclitic depending on syntactic conditions). See 4.1 for details.

- 2.2 Morphophonemic alternations inherited from Proto-Slavonic 1 Velar \sim alveo-palatal: /k g x sk zg/ \sim /č ž š št žd/- reflexes of the Proto-Slavonic first regressive palatalization of velars before front vowels and *j. Extremely common in Modern Bulgarian, the alternation is no longer conditioned solely by phonological factors. In inflection, it is limited to three categories: (1) a few masculine vocatives in /-e/: 'god' SG бот/bog, VOC боже/boze; (2) certain types of first-conjugation verbs: 'weep' 1 SG PRS плача/plača, 1 SG AOR плаках/plakax; (3) two anomalous neuter plurals: 'eye' SG око/oko, PL очи/oči and 'ear' SG yxo/uxo, PL уши/uši. In derivation, it occurs in diverse nominal and verbal formations; for example, мляко/mljako 'milk' : млечен/mlečen 'milky, of milk (м sG)', глух/glux 'deaf (м sG)' : глушея/glušeja '(I) grow deaf'.
- 2 Velar \sim dental: /k g x/ \sim /c z s/ \sim reflexes of the Proto-Slavonic second and third palatalizations of velars. The alternation is regular (though with exceptions) in the plural of masculine nouns with polysyllabic stems, for example, 'language, tongue' SG език/ezik, PL езици/ezici. It occurs in the plural of two common feminine nouns, 'hand, arm' SG ръка/ răka, PL ръце/răce and 'foot, leg' (dialectal) SG нога/noga, PL нозе/noze. In derivation it is uncommon; for example, малко/malko 'little, few': малцинство/malcinstvo 'minority'.

The interaction of the two preceding alternations creates a small number of alternations of /c z/ with /č ž/; for example, 'old man' SG crapeu/ starec : VOC старче/starče.

3 Jotation reflects the influence of PSI. * i on preceding dentals and labials (velars above):

labials: /p b v m/ ~ /pl' bl' vl' ml'/ dentals: /t d st s z n l r/ ~ /št žd št š ž n' l' r'/

Though once widespread in Old Bulgarian, these alternations are now extremely limited. In inflection, only the alternations /s z l r/ ~ /š ž l' r'/ are preserved in the paradigms of certain first-conjugation verbs: 'write' 1 SG PRS πμπα/ρίξα, 1 SG AOR πμααχ/ρίξαχ. In derivation, all of the alternations occur in isolated, unproductive formations. All examples of alternating labials, most of alternating /s z/ and many of /t d st/ are in borrowings from Russian or Russian Church Slavonic: 'send' 1 SG AOR PRFV пратих/pratix, 1 SG AOR IMPFV пращах/praštax; давя/davja '(I)

drown (ITR)', давление/davlenie 'pressure (from Russian or Russian Church Slavonic)'; коза/koza 'goat', кожа/koza 'skin'.

Rare examples of PSI. *kt/gt before front vowels, which develop like *ti. are included here; for example, mora/moga '(I) can', mout develop like '1,

This set of alternations was radically reshaped by two historical developments: (a) the simplification of sequences of labial and /l'/ to palatalized labials (the loss of 'epenthetic l' originally arising in Proto-Slavonic from labial + j) and (b) the substitution of /t' d' s' z'/ for /št žd š ž/, respectively. The result is the alternation of homorganic palatalized and nonconsonants, /p b f v m t d s z/ palatalized /p' b' f' v' m' t' d' ~ s' z'/, in place of the earlier alternations. The newer alternations are common in paradigms of certain first- and second-conjugation verbs; for example, 'drip' 1 SG PRS ка́пя/kápja, 1 SG AOR ка́пах/kápax; 'walk' 1 SG PRS хо́дя/хо́dja, 1 SG AOR хо́дих/хо́dix. They are also common in verbal derivation; for example, 'turn' 1 SG PRS PRFV извъртя/izvartja, 1 SG PRS IMPFV извъртявам/izvartiavam.

- 4 o ~ e: This alternation is a vestige of the Proto-Slavonic change of back vowels to front vowels after alveo-palatal consonants or the reflexes of the third palatalization of velars. Compare: градове/gradove 'cities', but броеве/broeve 'numbers' (SG брой/broj). The alternation is non-productive and lexically limited, and allows numerous alternatives; for example, змейове/zmejove ~ змееве/zmeeve 'dragons'.
- 5 Consonant truncation: Consonants are deleted before other consonants or at the end of a word. This alternation comprises a number of special cases. (a) /j/ is deleted in the inflection of first-conjugation verbs which show stem-final /j/ in the present tense; truncation here is regular and productive, though morphologically conditioned: 'drink' 1 SG PRS пия/ ріја, 1 SG AOR пих/ріх (but compare 2 PL IMP пийте/ріјtе). (b) Stemfinal /t d/ are regularly deleted before /l/ in the agrist participle of first-conjugation verbs: 'read' 1 SG PRS чета/četa, M SG AOR PART чел/ čėl. (c) /n s t/ alternate with Ø in singular and plural forms of some neuter nouns: 'time' SG време/vrėme, PL времена/vremena; 'taxi' SG такси/taksi, PL таксита/taksita. The alternations occur sporadically elsewhere in inflection and derivation: зная/znaja '(I) know', знак/znak 'sign'.
- 6 Vowel truncation: Stem-final vowels alternate with Ø in verbal inflection (the relevant stem elements are italicized in the following transliterated examples): 'walk' 1 SG PRS ходя/xodja, 1 SG AOR ходих/xodix; 'write' 1 SG PRS пиша/pisa, 1 SG AOR писах/pisax.
- $7/t/ \sim /s/$: This alternation is limited to a few derivationally related items, such as страдам/stradam '(I) suffer', страст/strast 'passion'.

8 Ablaut: Vestiges of Proto-Slavonic ablaut are preserved in derivation, particularly aspectual derivation; for example, бера́/berá '(I) gather', избор/izbor 'selection'; 'die' 1 SG PRS PRFV умра́/umrá, 1 SG PRS IMPFV уми́рам/umiram. A few examples occur in verbal inflection: 'gather' 1 SG PRS бера́/berá, 1 SG AOR бра́х/bráх.

2.3 Morphophonemic alternations resulting from changes after Proto-Slavonic

Here three types are distinguished with respect to the conditions of their application.

- 1 General, phonologically conditioned, optional alternations. In casual speech these apply across word boundaries as well as within words.
- (a) CC ~ C: Geminate consonants, which arise principally across morpheme boundaries (see above), alternate with their single counterparts: orryk/ottuk/ottuk ~ otuk/ 'from here'.
- (b) alveo-dental ~ alveo-palatal: /t d s z c/ ~ /č ž š ž č/, respectively, before /č ž š ž/: безжизнен/bezžiznen /bezžiznen ~ bežžiznen/, and with reduction of geminate /žž//bežiznen/ 'lifeless (M SG)'.
 (c) ST ~ S: /s z š ž/ before a word boundary or other consonant,
- (c) ST ~ S: /s z š ž/ before a word boundary or other consonant, /st zd št žd/ otherwise; for example, мощ/most /most ~ mos/ 'power', 'powerful' м SG мощен/mosten /mosten/ and F SG мощна/ mostna /mostna ~ mosta/.
- (d) E ~ J: unstressed /e/ and /i/ contiguous to a lower, usually stressed, vowel become /j/; thus, ματερμάπ/material /material ~ materjal/ 'material', δαιμά μ/baštά ὶ /baštái ~ baštáj/ 'her father' (= 'father her-DAT'). Similarly, /u o/ become [w]: Βοάπ/νοάl [voáł ~ vwáł] 'veil'. Sequences of consonant and [j]/[w] of this source are often pronounced as palatalized or labialized consonants: [mater'ál], [vwáł].
- 2 General, phonologically conditioned, but obligatory alternations.
- (a) Word-final devoicing: Word-final voiceless obstruents alternate with voiced obstruents before non-obstruents. Compare 'city' M SG град/grad /grat/ and PL градове/gradove /gradove/. While obstruents devoice before enclitics (except the definite article), they do not devoice at the end of prepositions; compare: град ли/grad li /gratli/'a city?' (compare градът/gradat/gradat/ 'the city') and под липите/pod lipite /podlipite/ 'under the lindens'.
- (b) Voicing assimilation in clusters: A voiced obstruent alternates with a voiceless obstruent before a voiceless obstruent; a voiceless obstruent alternates with a voiced obstruent before a voiced obstruent. For

example, 'sweet' M SG сладък/sladak /sladak/ ~ F SG сладка/ sladka /slatka/, cBat/svat /svat/ 'matchmaker' ~ cBat6a/svatba /svadba/ 'wedding'. The same alternation occurs between fully stressed words and clitics, and in colloquial speech between fully stressed words as well: οτ δάδα/οt baba /odbaba/ 'from grandmother', без това/bez tova /bestova/ 'without this (N SG)'. The alternation applies allophonically to /c x/; for example, uėtox ru/čėtox gi/čėtoyg'i/'(I) read (AOR) them'. Presumably inherited from Proto-Slavonic, the alternation became much more frequent with the loss of weak *ъ/ь.

/v v'/ followed by sonorants do not cause preceding voiceless obstruents to become voiced: сват/svat /svat/ 'matchmaker'; but от вдовицата/ot vdovicata /odvdovicata/ 'from the widow'. Between a voiceless obstruent and a sonorant /v v'/ are optionally voiceless: /svat ~ sfat/.

- (c) Velar ~ palatal: /k' g'/ before front vowels ~ /k g/ elsewhere (the rule applies allophonically to /x/); for instance, 'book' F SG книга/kniga /kniga/ ~ PL книги/knigi /knig'i/. It occurs across word boundaries in casual speech. The alternation arose after the change of PSI. *y to i, well after the velar palatalizations of Proto-Slavonic.

 (d) $C' \sim C$: Palatalized consonants before non-front vowels alternate with
- non-palatalized consonants elsewhere. This alternation is the result of relatively recent changes which eliminated palatalized consonants before front vowels, consonants, and word boundaries, as in 'land' F SG земя/zemja : PL земи/zemi : земна/zemna 'earthly, earthen (FSG)' : земляк/zemljak 'countryman'; 'blue' F SG синя/sinja: N SG синьо/ sin'o: M SG cин/sin: PL сини/sini 'blue'.
- (e) /i/ ~ 0: /j/ before back vowels, consonants or word boundaries alternates with Ø before front vowels; for example, M SG 'my' мой/moj versus PL мои/moi. This alternation is perhaps optional (Маслов/ Maslov 1981: 51).
- (f) /j ~ C'/: Root-initial /j/ alternates with palatalization of prefix-final consonant; thus, ям/jam '(I) eat' but обядвам/objadvam /ob'advam/ '(I) dine'.
- (g) /s/ ~ 0: /s/ alternates with 0 between /st st č š ž/ and /k/. This alternation occurs only in adjectives with the derivational suffix /-sk-/; for example, чешки/се́ški 'Czech (M SG)' from the noun /čex-/ + /-sk-/ with /x/ to /s/ (see 2.2).
- 3 Lexically restricted alternations, conditioned by phonological factors, morphological factors or a combination. All have exceptions.
- (a) C'à ~ Ce: Stressed /a/ preceded by a palatalized consonant alternates with /e/ (stressed or unstressed) preceded by a non-palatalized

consonant. The alternation is lexically limited to approximately 90–100 stems and several affixes (Scatton 1984, appendices 1 and 2). Otherwise it is conditioned by phonological factors: /C'a/ if the alternating syllable is stressed and not followed by (1) a palatalized or alveopalatal consonant; (2) a consonant cluster containing such a consonant; or (3) a syllable containing a front vowel; otherwise /Ce/ or /Ce/. Thus: 'blind' M SG cnán/sljap: PL cnénu/slèpi: cnénu/slèpčo 'blind man': 3acnená/zaslepja '(I) blind'. Exceptions are common: for instance, 'place', N SG Mácto/mjasto and PL места/ mesta, but related adjective местна/mestna 'local (F SG)'. The alternation is the result of the characteristic north-eastern Bulgarian treatment of PSl. *ě jat'.

- (b) Metathesis: As a result of the development of Proto-Slavonic liquid diphthongs with *Ъ/Ь (section 2.1), the position of the consonants /r l/ with respect to /a/ varies as a function of the following environment: /ra/ and /la/ before two consonants, /ar/ and /al/ before a consonant followed by a vowel. Lexically limited, the alternation is common in derivation, but infrequent in inflection; for example, върба/varba 'willow tree': Връбница/Vrabnica 'Palm Sunday'. In monosyllabic forms with more than one consonant after the liquid-vowel sequence, /a/ always follows the liquid: кръст/krast 'cross'. In other monosyllabic forms, both orders occur: гръм/gram 'thunder' versus вълк/valk 'wolf'. Exceptions are numerous: смъртна/smartna 'fatal (FSG)'. See Scatton (1984, appendix 3) for details.

 (c) Vowel ~ 6: In the inflectional and derivational patterns of many
- (c) Vowel ~ 6: In the inflectional and derivational patterns of many nominal and verbal forms, /ă/ and /e/ alternate with Ø. The alternation is the idiosyncratic property of many roots and affixes. For example, 'day' м SG ден/den: PL дни/dni; 'silver (ADJECTIVE)' м SG сребърен/srebăren: F SG сребърна/srebărna: сребърник/ srebărnik 'silver coin': сребро/srebro 'silver (N)'. The alternation is the result of the evolution of the PSI. *ъ/ь plus the apparent epenthesis of /ā/ in word-final clusters of consonant plus liquid (see section 2.1).

3 Morphology

3.1 Nominal morphology

3.1.1 Nominal categories Bulgarian nominal morphology includes the following categories:

1 Number, singular versus plural; masculine nouns use a separate count

form (Bulgarian бройна форма/brojna forma) with cardinal numerals.

- 2 **Definiteness**, expressed by a **definite article**, postposed to the first nominal constituent of definite noun phrases.
- 3 Case (extremely limited): Two classes of nouns and masculine singular adjectives have singular vocative forms (non-productive and limited). The masculine singular definite article contrasts subjective and objective forms (a distinction not strictly observed in colloquial Bulgarian). Personal pronouns and the masculine personal interrogative pronoun kom/koj 'who' (and other pronouns derived from it) have nominative, accusative and dative forms; non-clitic dative forms, however, are replaced by prepositional phrases (see 3.1.3).

Other case forms were common in the literary language in the past, particularly for masculine personal nouns; for example, NOM Владимир/Vladimir, DAT Владимиру/Vladimiru, ACC Владимира/Vladimira. They are now found only in regional, non-literary dialects (see section 6). For case vestiges in adverbial formations, see section 3.1.4.

4 Grammatical gender: masculine, feminine, neuter, reflected in the number suffixes (singular/plural) of nouns. In addition, gender and number are the basis for agreement of verbal and other nominal forms with nouns. Gender is natural only to the extent that certain nouns denoting humans, regardless of their inflectional patterns, are masculine or feminine for agreement purposes depending on their meaning; for example, although inflected like a feminine noun in /-a/, слуга/sluga '(male) servant' shows masculine agreement; likewise чичо/čičo 'uncle', although it is inflected like a neuter noun in /-o/. In addition, a small number of nouns with singulars in /-a/ which denote men or women show masculine or feminine agreement depending on the sex of their referents; for instance, пияница/ріјапіса 'drunkard'.

3.1.2 Noun morphology

The most important inflectional patterns of nouns are as follows:

1 Masculine singular /-0/

Monosyllabic	'son'	SG INDEF син sin	SG VOC си́не sine	PL INDEF синове sinove	PL COUNT сина sina
	'city'	гра́д gràd	гра́де grȧ̀de	градове gradovė	гра̀да gràda
Polysyllabic	'teacher'	учител učitel	учителю učitelju	учители učiteli	учителя učitelja

Vowel-zero alternations are common in this class: the vowel occurs in all singular forms and the count plural, zero in other plural forms: 'fool' SG rπynėų/glupėc, VOC rπynėųo/glupėco - PL rπynųų/glupci.

Vocative suffixes are /-e/, /-u/ (see above), or /-o/ ('man, husband' мъж/maž – мъжо/mažo; 'citizen' гражданин/graždanin – гражданино/graždanino). Their distribution depends on the stem-final consonant (Scatton 1984: 140-2).

/-ove/ is the regular plural suffix for monosyllabic stems. After /j/, it is sometimes replaced by /-eve/: 'tea' чай/čај – чаеве/čаеve. Often /-i/ replaces /-ove/, especially in borrowings: 'day' ден/den – дни/dni, 'fact' факт/fakt – факти/fakti (borrowed from Latin).

Several common monosyllabic nouns take other plural suffixes: /-išta/: 'path, road' път/pat – пътища/patišta; /-a/: 'brother' брат/brat – братя/bratja (with irregular palatalization of /t/); 'foot' крак/krak – крака/kraka; /-e/: 'man, husband' мъж/maž – мъже/mažė.

/-i/ is the regular plural suffix for polysyllabic stems; here stem-final velars generally become dentals: 'pupil' ученик/učenik – ученици/učenici. Exceptions are common, especially in borrowings: 'dinner jacket' (from English via Russian) смокинг/smoking – смокинги/smokingi. The suffix /-in-/ is lost in the plural: 'citizen' гражданин/graždanin – граждани/graždani. Rarely, /-ove/ occurs for expected /-i/: 'fire' отън/ogan – огньове/ogn'ove.

Alternative plural forms are common: 'sign' знак/znak – знаци/znaci ~ знакове/znakove.

The suffix of the plural count form is /-a/ - never stressed.

In vocative and count forms, stress falls on the same stem-syllable as in the singular. Monosyllabic stems show three possibilities in the plural:

Stem	ʻsign'	знак/znak	znakoве/znakove
Suffix initial	'chair'	стол/stol	столове/stolove
Suffix final	'city'	гра́д/grȧ́d	градове́/gradovė́

The first pattern is by far the most common. For polysyllabic stems, stress is fixed on the stem: 'pupil' ученик/učenik – ученици/učenici; an apparent shift to the plural suffix occurs with the loss of stressed suffixal vowels: 'fool' глупец/glupec – глупци/glupci.

2 Neuter nouns with SG /-o/ (/-e/), PL /-a/

	SG INDEF	PL INDEF
'place'	място/mjasto	места/mesta
'heart'	сърце/sarce	сърца/sărca
'doctrine'	yчение/učenie	учения/učenija

The singular suffix /-o/ may be /-e/ after stems terminating in palatalized,

palatal or, rarely, other consonants. Other suffixes are rare: /-i/ (two examples): 'eye' око/око – очи/осі, 'ear' ухо/ихо – уши/изі (both with velar to alveo-palatal); /-e/ (three examples): 'wing' крило/krilo – криле/krile (~ крила/krila), 'knee' коляно/koljano – колене/kolene (~ колена/kolena), 'shoulder' рамо/гато – рамене/гатепе (~ рамена/гатепа).

Stress is usually fixed on the stem ('doctrine' above) or the suffix ('heart'). Some nouns shift stress from stem to plural suffix, for instance 'place'.

A number of kinship terms and names for men have singulars in /-o/; they use the plural suffix /-ovci/; stress is fixed: 'grandfather' дядо/djado - дядовци/djadovci.

3 Neuter nouns with SG /-e/, PL /-eCa/

	SG INDEF	PL INDEF
'lamb'	а́гне/ágne	а́гнета/ágneta
'stool'	сто́лче/stólče	столчета/stolceta

This formation is regular for neuter nouns with the common diminutive suffix /-e/ or any of its variants, like /-če/. Stress is fixed. The suffix /-ta/ is productive for foreign borrowings terminating in /-i/ or /-u/ (unknown as singular suffixes in native nouns): 'taxi' τακτά/taksi – τακτάτα/taksita.

Seven nouns take stressed /-na/ instead of /-ta/: 'seed' ceмe/seme - семена/semena; also време/vreme 'time', име/ime 'name'.

Stressed /-sa/ occurs with two neuters in /-o/ or /-e/: 'sky, heaven' Hebe/nebe (archaic Hebo ~ Hebo/nebo ~ nebo) – Hebeca/nebesa, 'miracle' Ψύμο/čůdo – Ψυμεςα/čůdesa. A few other plurals of this type are archaic alternatives to regular formations; thus, 'speech' ςπόβο/slovo – ςποβα/slova and archaic ςποβεςα/slovesa.

4 Feminine nouns with SG /-a/

	SG INDEF	SG VOC	PL INDEF
'woman, wife'	жена	жено	жени
	žená	žėno	ženi
'land'	земя	земьо	земи
	zemjá	zėm′o	zemi

The vocative is regularly formed with /-o/ (see above) or /-e/ ('tsarina' царица/carica – царице/carice) (Scatton 1984: 140-2 for rules and exceptions). Stress in the vocative is always on the stem. The plural suffix is /-i/, before which velars do not become alveo-palatals. Two nouns take /-e/ with shift of velar to dental: 'arm, hand' ръка/raka – ръце/race and 'leg, foot' (dialectal) нога/noga – нозе/noze.

This pattern also includes some masculine personal nouns: '(male) servant' SG cπyrά/slugå, VOC cπýro/slúgo, PL cπyrή/slugi.

Vocative forms aside, the stress of nouns in /-a/ is fixed.

5 Feminine nouns with SG /-0/

	SG INDEF	PL INDEF
'song' 'bone'	песен/pesen кост/kost	песни/pesni кости/kosti
bone	KOCT/ KOSt	KOCTH/ KOSU

The plural suffix is /-i/. Vowel-zero alternations occur (section 2.3). Stress is fixed (except with the definite article: see below).

Many Bulgarian noun types are defective with respect to number. Some (singularia tantum) occur only in the singular: ypabhobecehoct/uravnovesenost 'equilibrium'; others (pluralia tantum) occur only in the plural: oчила/očila 'eye glasses'.

As the first constituent of a definite noun phrase, nouns carry the postposed definite article:

Singular:

```
masculine -0: -ът ~ -a (-ят ~ я)/-ăt ~ -a (-jat ~ at) (учителят/ učiteljat 'the teacher', глупецът/glupecăt 'the fool') feminine/masculine /-a/: -та/-ta (жената/ženata 'the woman, wife') feminine -0: -та/-ta (песента/pesenta 'the song')
     masculine /-o/, all neuter: -To/-to (мястото/mjastoto 'the place')
Plural:
     /-a/: -та/-ta (местата/mestata 'the places')
/-i/ or /-e/: -те/-te (жените/ženite 'the women, wives')
```

Notes:

- Masculine singular forms distinguish case: forms in /t/ nominative, forms without /t/ objective: τόμ ε προφέςορωτ/tόj ε profésorat 'he is the professor' versus сτάπτα нα προφέςορα/stájata na profésora 'the room of the professor'. In spoken usage, depending on regional dialect or idiolect, one of the two forms is generalized. With 1 some monosyllabic masculine nouns the stress unpredictably shifts to the article: rpaπ'bτ/gradat 'the city'. See section 2.1 for spelling. With feminine singular nouns in -Ø, stress is always on the article, which, additionally, is often pronounced [-ta]: [pesenta] ~ [pesenta].
- 2

Noun morphology has changed dramatically in the history of Bulgarian. Except for masculine and feminine vocative forms, all case forms were lost. The Proto-Slavonic nominative—accusative dual of o-stem masculine nouns was the source of masculine count forms. Otherwise, the dual was lost (except for vestiges: some feminine and neuter plurals in /-e/, some masculine plurals in /-a/, the plurals of 'eye' and 'ear' (examples above)). The definite article developed from the demonstrative pronoun PSI. *t-.

The declensional types of Modern Bulgarian continue the major patterns of Late Proto-Slavonic. Masculine and neuter stems with singulars in $/-\emptyset/$ and /-o/, respectively, are descendants of o-stem masculines and neuters. The plural suffix of monosyllabic masculines, /-ove/, preserves the nominative plural of u-stem masculines, otherwise absorbed by o-stems. The rare masculine plural suffix /- $\dot{e}/$, as in Mbkė/mažė, reflects nominative plural -be of masculine i-stems, otherwise absorbed by o-stem masculines. Feminine nouns in /-a/ continue Proto-Slavonic feminine \bar{a} -stems, feminine nouns in /-0/ -i-stems. Masculine personal nouns with singular in /-0/ are an innovation.

Vestiges of Proto-Slavonic 'hard' versus 'soft' inflectional patterns for o-and \bar{a} -stems are found in the vocative (though with redistribution of suffixes, including the extension of /-o/ to masculines), in the masculine plural suffix /-eve/ (for /-ove/), and in the neuter singular suffix /-e/ (for /-o/).

With one major exception, all minor classes of Proto-Slavonic nouns were either absorbed by major patterns or became non-productive and limited in number. As mentioned, u-stem and i-stem masculines merged with o-stem masculines. Most feminine \bar{u} -stems, developing singular forms in /-va/, merged with \bar{a} -stem feminines (u- \bar{b}) κ became feminines in /- θ / (κ p \bar{b} b/kr \bar{a} v 'blood'). Feminine \bar{i} -stems became feminines in /-a/ (κ pa \bar{b} b/kr \bar{a} v 'blood'). Masculine n-stems generally merged with masculines in /- θ / – with or without a change of suffix (π m \bar{a} b/plamak 'flame' versus π deh/den 'day'). The original suffix of 'stone' was more or less preserved in the plural while being replaced in the singular: κ am \bar{b} b/kamak – κ am \bar{b} b/kamani.

Neuter s-stems became neuters in /-o/ ('speech' слово/slovo - слова/slova); likewise for several neuter n-stems ('letter' писмо/pismo from OCS pismę). Several n-stems retain the old plural ('seed' cème/sème - семена/semena). Neuter nt-stems, preserved with plurals in /-ta/, are very common due to the productivity of diminutive suffixes in /-e/ and the use of /-ta/ with borrowings. Finally, two feminine r-stems, OCS mati 'mother' and dъšti 'daughter', became feminine nouns in /-a/: майка/ majka and пъщеря/dašterja (the latter with the vocative пъще/dašte).

3.1.3 Pronominal morphology

The personal pronouns of Bulgarian are shown in table 5.4. Ний/nij and вий/vij are normal in spoken Bulgarian. With decreasing consistency, second person plural forms are used for polite address of second person singular interlocutors. The clitic feminine dative singular is written with a grave accent to distinguish it from u/i 'and'.

Table 5.4 Personal pronouns

(a)	NOM	ACC		DAT		
()		Stressed	Clitic	Stressed	Clitic	
1 sg	åз	мене	ме	(мė́не)	ми	'I'
2 sg	ти́	тебе	те	(τۈбе)	ТИ	'you (SG)'
3 sg m	то́й	него	го	(нему)	му	'he/it'
N	то́	него	го	(нему)	му	'it'
F	тя́	нея	Я	(ней)	Й	'she/it'
1 PL	ни́е (ни́й)	нас	ни	(нам)	ни	'we'
2 PL	вие (вий)	вас	ви	(вам)	ви	'you (PL)'
3 PL	тė	тя́х	LN	(тя́м)	ИМ	'they'
REFL	_	себе си	ce	(се́бе си)	СИ	'-self'
(b)						
Ì śg	άz	mėne	me	(mėne)	mi	'I'
2 sg	ti	tėbe	te	(tebe)	ti	'you (SG)'
3 sg m	tój	nėgo	go	(nėmu)	mu	'he/it'
N	to	nėgo	go	(nėmu)	mu	'it'
F	tjå	nėja	ja	(nėj)	ì	'she/it'
1 PL	nie (nij)	nas	ni	(nam)	ni	'we'
2 PL	vie (vij)	vås	vi	(vam)	vi	'you (PL)'
3 PL	tė 🕽	tjåx	gi	(tjam)	im	'they'
		sebe si	se	(sebe si)	si	'-self'

Stressed dative forms, now archaic, are replaced by prepositional phrases of Ha/na + stressed accusatives. Stressed accusative forms serve as objects of prepositions and, with logical stress, as direct objects of transitive verbs. Clitic forms are normal as unmarked verbal complements, accusative as direct object, dative as indirect object:

Дадох него на нея./Dadox nego na neja.

Да́дох ѝ го./Dadox ì go.

Enclitic datives are the customary expression of possession (see section 4.9). 'Ethical' (modal) dative forms, usually clitics, are common in colloquial Bulgarian: той си пийва/tој si pijva 'he likes to take a drop' (= 'he self-DAT drinks-SEMELFACTIVE').

Personal pronouns are matched by **personal possessive pronouns**, inflected for gender and number like adjectives (see table 5.5). Like adjectives, possessive pronouns may carry the definite article (section 3.1.4). However, in first person singular, second person singular and reflexive

^{&#}x27;(I) gave it to her.'

^{&#}x27;(I) gave it to her.'

Table 5	.5	Personal	possessive	pronouns
---------	----	----------	------------	----------

	M SG	F SG	N SG	PL	
(a)					
1 sg	мо́й	мо́я	мое	мо́и	'my'
2 sg	тво́й	твоя	твое	тво́и	'your-sg'
3 sg m/n	не́гов	негова	негово	негови	'his/its'
F	неин	не́йна	не́йно	не́йни	'her/its'
1 PL	наш	на̀ша	наше	наши	'our'
2 PL	ваш	ваша	ваше	ваши	'your-PL'
3 PL	техен	тя́хна	тя́хно	техни	'their'
REFL	свой	своя	свое	свои	'-own'
(b)					-
ì śg	mòj	moja	mòe	mòi	'my'
2 sg	tvoj	tvoja	tvoe	tvòi	'your-sg'
3 sg m/n	nėgov	nėgova	nėgovo	nėgovi	'ȟis/its'
F	nėin	nėjna	nėjno	nėjni	'her'
1 PL	nàš	náša	naše	nàši	'our'
2 PL	våš	váša	váše	váši	'your-PL'
3 PL	tėxen	tjäxna	tjáxno	tėxni	'their'
REFL	svoj	svoja	svoe	svoi	'-own'

forms, /-ij-/ is not added before the masculine singular article: 'my' M SG DEF моят/mojat versus 'their' M SG DEF техният/texnijat.

Other pronominal forms, also inflected for gender and number, are demonstrative (table 5.6) and interrogative (table 5.7).

The masculine singular form of 'who' and all pronominal forms derived from it (below) have an oblique form, кого/kogo, used as direct object and object of prepositions. In colloquial Bulgarian it is often replaced by кой/koj. The archaic dative кому/komu is replaced by PREP на/па + кого/kogo.

With the exception of 'everyone, ...' (below), other pronominal forms are derived from interrogatives.

Relative pronouns add the suffix -то/-to: който/kojto 'who, which, that (M SG)', какъвто/kakavto 'which sort (M SG)'.

Negative pronouns add stressed ни-/ni-: никой/nikoj 'no one (M SG)', ничий/ničij 'no one's (M SG)', никакъв/nikakav 'no sort of (M SG)', нищо/ništo 'nothing (N SG)'.

Indefinite pronouns:

1 'some, a certain': stressed ня-/nja- + interrogative; here /a/ ~ with /e/; някой/njakoj 'someone (M SG)', някакъв/njakakav 'some sort

Table 5.6 Demonstrative pronouns

	M SG	F SG	N SG	PL
(a) 'this' 'that' 'such a'	то́зи/то́я о́нзи/о́ня такъ̀в	та́зи/та́я она́зи/она́я така́ва	това́/ту́й онова́/ону́й тако̀ва	те́зи/ти́я оне́зи/они́я таки́ва
(b) 'this' 'that' 'such a'	tozi/toja onzi/onja takav	tazi/taja onazi/onaja takava	tová/tůj onová/onůj takôva	tėzi/tija onėzi/onija takiva

- of (M SG)', нечий/nečij 'someone's (M SG)', нещо/nešto 'something (N SG)'.
- 2 'someone, so-and-so, such-and-such': èди-/èdi- + interrogative си/si, written as two words with secondary stress on the first constituent; èди-кой си/èdi-koj si 'so-and-so, a certain one (M SG)', èди-какъв си/èdi-kakav si 'such-and-such a, a certain (M SG)'.
- 3 'whoever, whatever, somebody or other, something or other, whatever sort of': interrogative + да è/da è, three words with secondary stress on the last; for example, кой да è/koj da è 'someone or other, anyone (M SG)', какво да è/kakvo da è 'whatever (N SG)'. Alternative forms replace the interrogative with a relative pronoun and add и/i: който и да è/kojto i da è.

Table 5.7 Interrogative pronouns

	M SG	F SG	N SG	PL
(a)				
'who, which' 'what'	ко́й	коя	коė какво/що	кой
'what sort of'	какъв	каква	какво	какви
'whose'	чий	кир	чие	чи й
(b)				
'who, which' 'what'	kój	kojå	koė kakvo/što	koi
'what sort of'	kakav	kakva	kakvó	kakvi
'whose'	čij	čijá	čiė	čii

4 'everyone, every single, each' M SG всеки/vseki, F SG всяка/vsjaka, N SG всяко/vsjako, PL всички/vsicki. Referring to persons, the masculine singular form of 'everyone ...' has accusative forms всекиго/vsekigo, всякого/vsjakogo; they are obsolescent in the colloquial language.

'everything' N SG всичко/vsičko

'every sort of' M SG всякакъв/vsjakakav, F SG всякаква/vsjakakva, N SG всякакво/vsjakakvo, PL всякакви/vsjakakvi

'all (of)' M SG всичкият/vsičkijat, F SG всичката/vsičkata N SG всичкото/vsičkoto, PL всички(те)/vsički(te)

See section 4, especially 4.7-4.9 for pronominal usage.

3.1.4 Adjectival morphology

Adjectives are inflected for number and, in the singular, gender; there is a masculine singular vocative form:

	'new'	'goat's'	'fraternal'
M SG	но́в∕nóv	ко́зи/kózi	братски/bratski
M SG VOC	но́ви/nóvi	ко́зи/kózi	братски/bratski
F SG	но́ва/nóva	ко́зя/kózja	братска/bratska
N SG	ново/novo	ко́зе/kóze	братско/bratsko
PL	но́ви/nóvi	ко́зи/kózi	братски/bratski

The usual masculine singular suffix is /-Ø/. The suffix /-i/ is limited to adjectives in /-sk-/, a small number of possessive adjectives, and a few Russian loans in /-ov-/ (брегови/bregovi 'shore (ADJECTIVE)'). Stems with vowel-zero alternations have a vowel in the masculine singular before /-Ø/, otherwise zero: 'good' м SG добър/dobar - F SG добра/dobra.

The regular neuter suffix /-o/ may be /-e/, as in κόσε/köze (see section 2.2).

Most adjectives are stem-stressed. End-stress is limited to Russian borrowings with /-ov-/ (see above) and two native stems: 'good' (above), 'alone' M SG cam/sam (F SG cama/sama).

A few colloquial adjectives borrowed from Turkish and, more recently, other languages are not inflected: сербез човек/serbez čovek 'bold man', сербез жени/serbez ženi 'bold women' (see section 5.3).

Masculine singular forms in /-i/ continue definite Proto-Slavonic forms; others continue indefinite forms. All earlier non-nominative case forms, the dual and, in the plural, gender were lost. Except for irregular neuter singular forms in /-e/, all trace of the soft adjectival paradigms of Proto-Slavonic was lost. The number of adjectives with palatalized stem-final consonants is limited to a few possessive adjectives (козя/kozja 'goat's (FSG)'), and one non-derived adjective 'blue' м SG син/sin, F SG синя/sinja, N SG синьо/sin'o, PL сини/sini.

Table 5.8 Cardinal numerals

(а)	'11' единадесет '12' дванадесет '13' тринадесет '14' четиринадесет '15' петнадесет '16' шестнадесет '17' седемнадесет '18' осемнадесет '19' деветнадесет 'million' милион	'50' петдесет '60' шестдесет '70' седемдесет	'300' '400' '500' '600' '700' '800' '900' '1,000'	шестстотин седемстотин осемстотин деветстотин
(b) '1' edin '2' dvå '3' tri '4' četiri '5' pėt '6' šėst '7' sėdem '8' osem '9' dėvet '10' dėset	'11' edinadeset '12' dvanadeset '13' trinadeset '14' četirinadeset '15' petnadeset '16' šestnadeset '17' sedemnadeset '18' osemnadeset '19' devetnadeset 'million' milion	'20' dvådeset '30' trideset '40' četirideset '50' petdeset '60' šestdeset '70' sedemdeset '80' osemdeset '90' devetdeset '100' sto 'thousand million'	'300' '400' '500' '600' '700' '800' '900' '1,000'	čétiristòtin pétstòtin séststòtin sédemstòtin osemstòtin

Comparative adjectives are formed with preposed stressed по-/ро-: м SG по-нов/ро-nov 'newer'. Superlatives use най-/naj-: м SG най-нов/naj-nov 'newest'. Note the regular 'double' stress in both formations.

The synthetic comparative forms of Proto-Slavonic were lost – with vestiges in висш/visš 'higher, superior', низш(и)/nizš(i) 'lower, inferior', старши/starši 'elder, senior', младши/mladši 'junior'.

As the first constituent of definite noun phrases, adjectives carry the definite article: M SG /-ă(t)/, F SG /-ta/, N SG /-to/, PL /-te/. Before the masculine singular definite article, the adjectival stem is extended with /ij/: 'the new' M SG новия(т)/novija(t), F SG новата/novata, N SG новото/ novoto, PL новите/novite. See section 2.1 for spelling of masculine singular forms. Forms with /t/ are subjective, those without it are objective (section 3.1.2). Stress remains unchanged with the definite article.

Certain adjectival forms function productively as adverbs: (1) masculine singular (or plural) forms of adjectives in /-sk-/ (братски/bratski 'fraternal' and adverb 'fraternally'); (2) neuter singular forms of qualitative adjectives with masculine singular in /-Ø/ (хубаво/хивачо 'nice' and adverb 'nicely');

(3) colloquially, definite feminine singular forms (здравата/zdravata 'healthy-the' and adverb 'soundly'). The comparison of adverbs follows rules for the comparison of adjectives: πό-χύδαβο/ρό-χώβανο 'more nicelv'.

Non-productive, lexicalized adverbial formations - from noun and adjectival bases – often preserve old case distinctions; thus, locative долу/dolu 'downward'; instrumental тихом/tixom 'quietly'. Similarly in frozen prepositional phrases: снощи/snosti 'last night' (preposition c/s 'from' and genitive singular of нощ/nost 'night').

Numeral morphology

Cardinal numerals (see table 5.8): spoken and sometimes written Bulgarian use /-najset/ for /-nadeset/ ('11' единайсет/edinajset), and in '20', '30', '40', '60' /-jset/ for /-deset/ ('20' двайсет/dvajset).

Other cardinals are made up of the forms in table 5.8. Here 'thousand' has the plural хиляди/xiljadi (with shifted accent), and 'million' and 'thousand million' have the count forms милиона/miliona, and милиарда/miliarda. For example, '36,620,105' триста и шест милиона, шестстотин и двадесет хиляди, сто и пет/trista i šest miliona, šeststotin i dvadeset xiljadi, sto i pet. Note the regular use of и/i 'and' in each compound term.

'1' is inflected like an adjective: една книга/edna kniga 'one book'. Its plural is used with pluralia tantum: едни клещи/edni klesti 'one (set of) pliers'. Increasingly '1' functions as an indefinite article 'a(n), some': един човек/edin čovek 'a person'.

'2' два/dva is used with masculine nouns, две/dve with neuters and feminines: два града/dva grada 'two cities', две жени/dve ženi 'two women (F)', две места/dve mesta 'two places (N)'.

There are special forms of '2'-'6' for masculine personal nouns (and

groups of male and female persons):

- '2' двама/dvama
- '3' трима/trima
- '4' четирима/četirima ~ четирма/četirma
- '5' петима/petima
- '6' шестима/šestima

For example, двама ученици/dvama učenici 'two pupils'.

Cardinal numerals may be definite. In this respect '1' is adjectival: 'the one' м sg единият/edinijat, F sg едната/ednata. 'Million', 'thousand million' and all cardinals in /-a/ follow rules for nouns: 'the million' милионът/milionat, 'the two (M)' двата/dvata. Remaining cardinals use /-te/, always stressed except with '2 (F/N)' and '3': 'the two (F/N)' двете/ dvete. 'the 700' селемстотинте/sedemstotinte.

Ordinal numerals are adjectives formed from cardinals with characteristic adjectival gender/number suffixes. Masculine singular ordinals up to and including '90' use /-i/, those from '100' on use /-Ø/; пети/peti '5th (M SG)', стотен/stoten '100th (M SG)'. Note:

- 1 Suppletive forms for:
 - '1st' първи/parvi (~ пръв/prav) '2nd' втори/vtori

 - '3rd' трети/treti
 - '4th' четвърти/četvarti
- '100th' is /stoten/; /e/ alternates with \emptyset and /o/ is stressed: 2 деветстотно/devetstotno '900th (N SG)'.
- 'Thousandth', 'millionth' and 'thousand-millionth' use the adjectival 3 suffix /-en-/; /e/ alternates with Ø: хиляден/xiljaden 'thousandth (M SG)'.
- Accent shifts take place in the formation of '9th', '10th', '40th': девети/deveti, десети/deseti, четиридесети/četirideseti. 4
- Stem final vowels are lost in '7th' седми/sedmi and '8th' осми/оsmi. 5

In compound phrases only the final elements are ordinals: триста петдесет и втори/trista petdeset i vtori '152nd (м sG)'.

Definite ordinal numerals are formed as definite adjectives: вторият/ vtoriiat 'the second (M SG)', стотната/stotnata 'the hundredth (F SG)'.

3.2 Verbal morphology

3.2.1 Verbal categories

Finite Bulgarian verbal forms are simple or compound. Simple forms convey person – first, second, third – and number – singular, plural. In addition, compound forms using participles show gender in the singular: masculine, feminine, neuter.

There are three tenses in the indicative mood - present, past, future which, combined with other categories, occur in nine formations:

1 Present, temporally unmarked, a simple form made up of verbal stem plus complex suffix - /e/ or /i/ (both \emptyset in first person singular and third person plural) or /a/, plus person/number marker:

	Singular	Plural
First person	/-a~ -m/	/-m ~ -me/
Second person	/ - š/	/-te/
Third person	/-Ø/	/-ăt/

2 Past imperfect, a simple form, expressing actions contemporaneous with

or subordinate to other past actions; made up of the present-tense stem plus a complex suffix consisting of a vowel $/e \sim \dot{a}/$ followed by:

	Singular	Plural
First person	/-x/	/-xme/
Second person	/-še/	/-xte/
Third person	/-še/	/-xa/

3 Past aorist, a simple form, relating a temporally independent, concrete past action; the aorist stem plus:

	Singular	Plural
First person	/-x/	/-xme/
Second person	/-Ø/	/-xte/
Third person	/-Ø/	/-xa/

4 Future, a compound form:

invariant proclitic auxiliary ще/šte + present tense

Negated future forms, and other forms involving the future (see below), use the neuter third person singular auxiliary няма да/пјата da ('has-not to'); note the colloquial alternative не ще/ne štė.

5 **Present perfect**, a compound form expressing an action completed in the past but relevant for or related to the present:

present tense 'be' + aorist past active participle

6 Past perfect, a compound form expressing an action completed in the past relative to another past action:

past tense 'be' + aorist past active participle

7 Future perfect, a compound form expressing an action to be completed in the future prior to another future action:

future tense 'be' + aorist past active participle

8 Past future, a compound form expressing an action to be completed in the past but future with respect to another past action; commonly used in conditional constructions (see section 4.5):

imperfect past of щя/štjå '(I) will, want' + да/da + present tense

9 Past future perfect, a compound form conveying a past action which is

Table 5.9 Indicative forms

	Non-perfect		Per	fect
	Non-future	Future	Non-future	Future
(a) Non-past Past Aorist Imperfect	пи́ша пи́сах пи́шех	ще пи́ша щя́х да пи́ша	писал съм бях писал	ще съм писал щях да съм писал
(b) Non-past Past Aorist Imperfect	piša pisax pišex	šte piša štjax da piša	pisal săm bjáx pisal	šte sām pisal štjáx da sām pisal

future with respect to a past action, which itself is prior to another past action; rare, commonly replaced by the past future:

past future of щя/štja '(I) will, want' + aorist past active participle

Thus 'write (3 SG M IMPFV)' has the paradigm shown in table 5.9.

Bulgarian verbs also express aspect: perfective verbs mark the completion of the action of the verb, imperfective verbs are unmarked. Most Bulgarian verbs have perfective-imperfective pairs: 1 SG PRS IMPFV пиша/piša '(I) write' – PRFV напиша/napiša '(I) write down, complete writing'. 'Bi-aspectual' verbs – verbs with one stem for both aspects – are common; most are borrowings: тренирам/treniram '(I) train (IMPFV/PRFV)' (from English) (see section 5.3).

Morphological processes related to the creation of aspectual pairs begin with non-prefixed imperfective verbs (see section 3.3.3), from which perfective stems are formed by suffixation or prefixation. Besides changing the aspect, perfectivizing prefixes often add their own meaning: допиша/dopiša '(I) finish writing (PRFV)' versus подпиша/podpiša '(I) sign' (for lists and definitions of prefixes see Граматика/Gramatika 2: 217ff.). The single perfectivizing suffix /-n-/ imparts semelfactive meaning to the base: падна/padna '(I) fall (PRFV SEMELFACTIVE)' (< падам/padam '(I) fall (IMPFV)').

'Secondary' imperfectives, all third conjugation (see below), are formed from perfectives by suffixation. While this process allows many alternative forms and shows random consonantal and vocalic alternations, there are only two imperfectivizing suffixes. The non-productive suffix /-a-/ occurs with a limited number of stems; 'speak' 1 SG PRS IMPFV robops/govorja >

'converse' 1 SG PRS PRFV разговоря/razgovorja > 1 SG PRS IMPFV разговарям/razgovarjam. The **productive** suffix is /-(a)va-/ (unstressed initial /a/ is deleted): 1 SG PRS IMPFV 'count' броя/broja' > 'enumerate' 1 SG PRS PRFV изброя/izbroja' > 1 SG PRS IMPFV изброявам/izbrojavam; 'write' 1 SG PRS IMPFV пиша/ріšа > 'copy' 1 SG PRS PRFV препиша/ргеріšа > 1 SG PRS IMPFV преписвам/ргерізvam. See Scatton (1984: 285-310) for details.

The Proto-Slavonic aspectual distinction between determinate and

indeterminate verbs of motion is entirely lost in Bulgarian.

Aspect is subject to some constraints. Only imperfective verbs have present active participles, gerunds, verbal nouns in /-ne/ and negative imperatives. Only imperfective verbs can be complements of verbs meaning 'begin', 'continue' or 'end'. Present perfective verbs are rare in independent clauses. Imperfect forms are most frequently imperfective aspect, and aorists perfective. Perfective imperfects and imperfective aorists are possible: perfective imperfects for usually repeated series of completed actions presented as subordinate ('backgrounded') with respect to other, 'major' past actions; imperfective aorists for 'major' past events the completion of which is not relevant for the narration. For discussion and examples see Scatton (1984: 318-32) and Lindstedt (1985).

Imperatives are simple or compound. Simple forms are second person

singular or plural; there are compound forms for all persons and numbers. Simple forms use the suffixes SG /-i/ and PL /-ete/; the suffix-initial vowel is /j/ with third-conjugation verbs and with verbs of other conjugations whose present stems terminate in /j/; compare 'write' 2 SG пиши/ріšі, 2 PL пишете/pišete; 'look' 2 sg гледай/gledaj, 2 PL гледайте/gledajte. The vowel is lost unpredictably in a few other verbs: влез/vlez 'enter (2 SG)'. See section 4.2 for compound imperatives.

The conditional is a compound form using the agrist past tense of the

stem /bi-/ 'be' (used only here) plus aorist past active participle: 'write' 1 SG M δίαχ πάςαπ/bix pisal '(I) could/should/would write/have written'. See section 4.5 for conditionals using the indicative.

There are several passive formations:

- 1 Any tense of 'be' + past passive participle of transitive verbs: писмото беше написано/pismoto bese napisano 'the letter was written'.
- 2 Finite forms of transitive verbs with the accusative reflexive pronoun ce/se: писмото се пише/pismoto se piše 'the letter is (being)
- 3 'Impersonal' (subjectless) third person plural finite forms of transitive verbs: там продават мляко/tam prodavat mljako 'milk is sold there' (= 'there sell-3 PL milk').

These formations have slightly different semantic and grammatical nuances (Граматика/Gramatika 2: 245ff.). A rare 'impersonal passive' uses 'be' and neuter singular past passive participles of transitive and intransitive verbs: по травата е ходено/ро travata e xodeno 'someone (or something) has walked on the grass' (= 'on grass-the is walked'; Γραματμκα/ Gramatika 2: 381).

Reflexive verbs, which carry the clitic accusative reflexive pronoun, have various functions:

- 1 Many common intransitive verbs are reflexive. Some occur only reflexively: сме́я се/sméja se '(I) laugh'; others are derived from non-reflexive transitive verbs: бия/bija '(I) beat (TR)' > бия се/bija se '(I) fight (ITR)'.
 2 Reflexive verbs derived from non-reflexive transitive verbs occur in
- passive constructions (see above).
- Reflexive forms of transitive and intransitive verbs are used in third person singular 'impersonal' constructions: (a) to express desire, спи ми ce/spi mi se 'I want to sleep' (= 'sleep-3 sg me-DAT self-ACC'); (b) to express permission, тук не се пуши/tůk ne se půši 'no smoking here' (= 'here not self-ACC smoke-3 sG').

For reciprocal and genuinely reflexive uses of reflexive verbs see section 4.8.

Indicative forms relate events personally witnessed or otherwise assumed to be true by speakers. For events not witnessed or known only through hearsay, Bulgarian uses so-called 'renarrated' forms (Chvany 1988). Every indicative form has a renarrated form; however, each renarrated tense, except one, corresponds to two indicative tenses:

Indicative	Renarrated	
present/imperfect	present	present 'be' + imperfect past participle
aorist	aorist	present 'be' + aorist past participle
present/past perfect	perfect	perfect 'be' + aorist past participle
(past) future	future	perfect 'will, want' + /da/ + present
(past) future perfect	future perfect	perfect 'will, want' + /da/ + perfect

In addition, the present auxiliary 'be' is deleted in all **third-person** forms. Thus, the renarrated paradigm for 'write (3 SG M IMPFV)' shown in table 5.10. Compare **indicative** той написа писмото/toj napisa pismoto 'he

s. 10. Compare indicative той написа писмото/toj napisa pismoto 'he wrote the letter (and I saw him do it)' versus renarrated той написал писмото/toj napisal pismoto '(it is said that) he wrote the letter'.

First-person renarrated forms are uncommon; they convey a sense of surprise or denial: пишел съм роман/pišel sam roman 'I'm writing a novel!?' (= 'it's claimed that I'm writing a novel'). Emphatic present, aorist, future and future perfect renarrated forms replace the present-tense auxiliary 'be' with its present perfect form, while dropping the third-person

Table 5.1	10	Renarrated	forms
-----------	----	------------	-------

	Non-perfect		Per	fect
	Non-future	. Future	Non-future	Future
(a) Non-past Aorist	пишел писал	щя́л да пѝша	бил писал	щя́л да е пи́сал
(b) Non-past Aorist	pišel pisal	štjál da piša	bil pisal	štjál da e pisal

present auxiliary: renarrated пишел/pišel – emphatic renarrated бил пишел/bil pišel '(he) writes' (Граматика/Gramatika 2: 361).

Some grammars describe another renarrated form, one which relates facts not personally witnessed by speakers, but **presumed** by them to be true. **Presumptive** forms use the present tense of 'be' plus a past participle, either aorist or imperfect (only aorist participles are used in indicative forms): той е пишел писмото/toj е pišel pismoto 'he (presumably) wrote the letter'. Largely identical to other indicative or renarrated forms, this category is controversial (Scatton 1984: 332-3).

Non-finite verbal forms are the following:

The late Proto-Slavonic **infinitive** and **supine** have been replaced by phrases with да/da + present tense: искам да пиша/iskam da piša '(I) want to write'. A vestigial **infinitive** of very limited use is identical to the second-third person singular agrist (see section 4.5).

Bulgarian lost the present active, present passive and the 'first', non-resultative past participles of Proto-Slavonic. On the other hand, it preserved two other participles and innovated two more.

A present active participle is formed from imperfective present stems with the suffixes /-ašt- ~ -ešt-/: M SG пишещ/pišešt 'writing'. A strictly literary form devised in the late nineteenth century on Russian and Church Slavonic models, it is only used attributively.

The imperfect past active participle is formed from imperfect present stems with /-el- ~ -al-/: м sg пишел/pišel 'wrote'. Used only in renarrated forms, it is a Bulgarian innovation.

The aorist past active participle is formed from aorist stems with the suffix /-l-/: M SG πμαπ/pisal 'wrote'. The direct descendant of the 'second',

resultative past active participle of Proto-Slavonic, it is used in compound verbal forms and, occasionally, attributively.

The past passive participle is formed from aorist stems with the suffixes /-(e)n- ~ -t-/: M SG писан/pisan 'written'. Continuing the past passive participle of Proto-Slavonic, it is used predicatively in passive formations and attributively.

Based on western Bulgarian forms which continue the Proto-Slavonic present active participle, the Bulgarian **gerund** (verbal adverb) is formed from imperfective present stems with the suffix -(e)jki: пишейки/ріšеjki '(while) writing'. It relates an action contemporaneous with, but subordinate to, the main verb. It is a strictly literary form. For more on participles and gerunds see section 4.5.

A neuter singular **noun**, denoting the action of the verb, is formed with /-(e)ne/ from imperfective stems: nicane/pisane 'writing'.

3.2.2 Conjugation

Traditional grammars distinguish three conjugations on the basis of the initial vowels of present-tense suffixes:

First conjugation	/e/	пи́ш-е-ш/рі́š-е-š	'(you-SG) write'
Second conjugation	/i/	хо́д-и-ш/xöd-i-š	'(you-sg) go, walk'
Third conjugation	/a/	и́м-а-ш/im-a-š	'(you-sg) have'

Various subclasses of the first and second conjugations reflect the relationship between the stems of the **aorist past** and **present** tenses. For example, the first subclass of the first conjugation makes its aorist stem by adding /e/ (2/3 sG) or /o/ (elsewhere) to the present stem. Table 5.11 gives examples of major subclasses and their characteristic alternations. Double stresses indicate alternative pronunciations. See Παμιοβ/Pašov (1966) and Γραματικα/Gramatika (2: 304ff.) for exhaustive lists.

Other verbal forms belong to one or the other of two 'systems', depending on which of the two stems is the basis for their formation:

Present system
present tense
imperfect past tense
imperative
present active participle
imperfect past participle
gerund

Aorist system aorist past tense aorist past participle past passive participle infinitive (vestigial) verbal noun

The first and second conjugations continue thematic patterns of Late Proto-Slavonic – with characteristic Bulgarian changes: loss of dual, loss of

Table 5.11 Bulgarian conjugational patterns

chto cisti чета чета чета чето 'read' ido iti ида идох 'go' gebo greti река рече рекох 'say' zivo ziti ла-сьпо ла-сфи ла-фи гина 'perish' ла-сьпо ла-сфи за-чена за-чена за-чена за-чена за-чена усопсеіче' комо комо кома кова сова сова 'call' 'become' веро рьгаті lc кажа кажа кова кова кова кова кова 'call' веро рьзаті lc кажа кажа кажа кажа 'write' ll мра мра мра 'die' 'write' ll кажа кажа кажа кажа		Church Slav 1 SG PRES			rn Bulgarian 1 SG PRES	3 SG PRES	1 sg aor	
vedo vesti до-веда до-веде четох 'read' ido iti ида иде идох 'go' reko resti река рече рекох 'say' say' perish' за-ченах 'perish' за-ченах 'perish' за-ченах 'go' за-ченах 'begin' по-чна по-ч	(a) I			la				
vedo vesti до-веда до-веда по-веда vere verox 'bring' (= 'li ido iti vera vere verox 'read' ido vera vera vera vera vera vera vera vera		neso	nesti		до-неса	до-несе	до-несох	
ido iti reko rešti pekox 'go' rešti pekox 'say' say' peke pekox 'say' say' resti peko greti zivo ziti na-četo na-četi mbro na-četi mbro mreti stano stati kovo kovati zovo zъvati bero bьгаті ll gybno gybati mino minati ll kažo kazati pišo pьзаті ll kažo kazati pišo pьзаті ll kažo kryi kryi kryi bijo biti nek ryjo kryti bijo biti nek ryjo petti umėjo umėti dėlajo darovati sėjo sėjati ll moljo moliti xoždo xoditi lla mona mona meti dėlajo darovati sėjo sъpati lla mona mona meti dėlajo terpetti ležo ležati nek mona mona mona mona mona mona mona mona		vedo			до-веда			'bring' (= 'lead'
reko rešti grebo greti peka pevė pėkox 'say' perish'		čьtǫ	čisti		чета	чете	четох	'read'
grebo greti ziti — гина пине пинах 'perish' winha мина мине минах 'go' sa-чене за-ченах 'conceive' hegin' по-чна по-чна по-чнах 'begin' rpeба греба гребах 'row' craha crahax 'become' kowa kowati zovo zъvati bero bьгаti вето bьгаti вето bьгаti вето вето вытаti вето вето вето вытаti вето вето вето вытаti вытаti вето вытаti вина вытаti вина вытаti вина вытаti вина вытаti вина вытаti вытаti вытаti вина вытаti вытаt		idǫ				иде	и́дох	ʻgoʻ
Živo Žiti мина мина за-чена за-ченах 'conceive' па-čьпо па-čьрі па-čері по-чна по-чна по-чнах 'begin' преба греба гребах 'row' 'begin' гребах гребах 'forge' 'conceive' кома кова ковах 'forge' зова зова зовах 'sall' шта піма мра мрах 'die' пра кажа кажа кажа кажа кажа 'know' греба куті кра		reko			• .	• .		'say'
па-съно па-сърті па-сърті па-съно па-сърті па-съно па-сърті па-сърті па-чна по-чна по			· ·	/ -				
ma-сьпо па-сеті што штеті stano stati kovo kovati zovo zъvati bero bьгаti ll gybno minati ll gybno minati ll gybno minati ll kažo kazati pišo pьsati ll kažo kazati pišo pьsati ll kažo kryjo kryti bijo biti pojo pětti umějo uměti dělajo dělati wwine meljo moliti xoždo xoditi lt tьгріjo tьгрěti ležo ležati other III moljo moliti xoždo xoditi lt tьгріjo tьгрěti ležo ležati other limatь imati limati		živo	žiti —	\ / <i>r</i> •				
mbro mrěti stano stati kovo kovati zovo z-bvati bero bьrati le yo bъrati le yo biti pojo peti pojo wměti dělajo dělati www. yo be able yo beati pojo yo peti pojo wměti dělajo dělati www. yo beati pojo yo beti pojo wměti dělajo dělati www. ww. ww. ww. ww. ww. ww. ww. ww. w		กล-čъпо	na-četi	$\setminus //$				
stano stati kovo kovati zovo zъvati bero brati bero brati bero brati mino minati ll sažo kazati pišo pьsati ll kažo kazati pišo pьsati ll sajo znati ll sajo znati le znajo znati ll superiori se zovo zbati lo superiori se zbati lo superiori superio		na cony		X				
kovo kovati zovo zъvati bero bьгаti III gybno gybati mino minati III kažo kazati pišo pьsati Id мра мре мрах 'aii' ile мра мре мрах 'die' Id мра мре мрах 'hear' kryjo kryti bijo biti pojo pěti uměti dělajo dělati wmějo uměti darujo darovati sějo sějati IV moljo moliti xoždo xoditi III III III III III III III		тьго	mrěti —	// >				
zovo zъvati bero bьгаti bеro bьгаti bеro bьгаti bеro bьгаti bеro bьгаti bеro bьгаti bьго bьго bьгаti bьго bьгаti bьго bьгаti bьго bьгаti bьго bьгаti bьго bьго bьгаti bьгаti bьго bьгаti bьго bьгаti bьг				//				
bero bьгаti gybno gybati mino minati III kažo kazati pišo pьsati Id мра мре мрях 'die' Id мра мре мрях 'die' znajo znati If чуя чуе чух 'hear' кујо куті бия бия бия бия бия бия ген пях 'sing' умея уме умях 'be able' мивея живея живе живях 'live' сея сее сях (сеях) 'sow' meljo mlěti darujo dělati IV moljo moliti хоždo хоditi III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo sejol sejati III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'beg' ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'beg' ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'beg' ходя ходи ходих 'walk' III moljo sejol sejati III moljo moliti ходя ходи ходих 'beg' ходя ходи ходих 'walk' III moljo sejol s		kovo	11	//		кове		
III Ražo kazati pišo pьsati Ic kama kame kajak 'say' minua niume niuciak 'write' Id mpa mpe mpm 'die'		zovo	zъvati \\	//	зова	зове	зовах	'call'
gybno gybati mino minati III kažo kazati pišo pьsati Id мра мре мрях 'die' Id мра мре мрях 'die' Id мра мре мрях 'die' If чуя чуе чух 'hear' kryjo kryti kryis kryis kryis kryis bijo biti пея пее пях 'sing' yme's yme'		bero	bьrati 📈	7	бера	бере	бра҅х	'gather'
mino minati la kažo kazati pišo pьsati la kažo kazati pišo pьsati la kažo kazati pišo pьsati la kažo pъsati la kažo pьsati la kažo pъsati la kažo pisati la	II		- λλ					
III kažo kazati pišo psati Id kaža kaža kaža kažaža 'say' niuma niume niucax 'write' Id Mpå Mpė Mpя́x 'die' le Jahan Jahae Jahan 'know' If vyn vye vyx 'hear' kryjo kryti bijo biti pojo pěti umějo uměti dělajo dělati meljo mlěti darujo darovati sejo sějati IV moljo moliti xoždo xoditi IIa Monn Monn Monn Monn Monn 'grind' IIa Monn Monn Monn Monn 'grind' IIa Monn Monn Monn Monn 'beg' xody xoditi IIb Tъpnя тъpпи тъpпях 'tolerate' neжа лежи лежах 'recline' other cnя cnи cnax 'sleep' III imamь imati III Mam Monn Monn Monn 'sleep' neжа лежи лежах 'recline' other cnя cnи cnax 'sleep' III imamь imati III Mam Monn Monn Monn Monn Monn Monn Monn Mon				1				
kažo kazati pišo pьsati Id niuma niume niucax 'say' mite' Id mpa mpe mpax 'die' Ie зная знае знаях 'know' If чуя чуе чух 'hear' kpis kpis kpis kpis kpis kpis kpis kpis		mino	minati —/	1				
ріšǫ рьзаti Id мра́ мре́ мря́х 'die' Ie зна́я зна́е зная́х 'know' If сијо čuti кгујо kryti ріјо ре́ti ројо ре́ti пе́я пе́е пя́х 'sing' уме́я уме́е умя́х 'be able' ите́јо uměti darujo darovati séjo séjati IV тојо тојо тојт търпи търпя́х 'tolerate' recline' ите́јо tьгре́ti ре́до sъраti штать ітаtі пиша пише писа́х 'write' мра́ мре́ мря́х 'die' чу́я чу́е чу́х 'hear' 'know' кри́я кри́е кри́х 'bide' би́я би́е би́х 'beat' пе́я пе́е пя́х 'sing' уме́я уме́е умя́х 'be able' отмет ме́ля ме́ле мля́х 'grind' Па мо́ля мо́ли мо́ли́х 'beg' хо́ди хо́ди́х 'walk' Пропря́ търпи́ търпя́х 'tolerate' recline' отмам и́ма и́мах 'have' дя́лам дя́ла дя́лах/дела́х 'carve'	Ш		1	\ lc				
Id le мра мре мрях 'die' Ie зная знае знаях 'know' If чу́я чу́е чу́х 'hear' kryjo kryti кри́я кри́е кри́х 'hide' bijo biti ројо реті пе́я пе́е пяҳ 'sing' уме́я уме́е умяҳ 'be able' delajo delati живе́я се́е сяҳ (се́яҳ) 'sow' IV moljo moliti ходя ходи ходих 'beg' ходя ходи ходих 'beg' ходя ходи ходих 'walk' IIb търпя търпи търпях 'tolerate' лежа лежи лежа пежа пе́е пяҳ 'sleep' III шать imati шать спаҳ 'sleep' III шать imati шать паҳ шая паҳ паҳ 'have' дяҳам усатем 'die' 'know' 'know' 'know' 'hear' 'know' 'hear' 'know' 'beat' 'poҳ θαҳ γνи γνи γν		•		11				•
znajo znati le зная знае знаях 'know' lf vyя чуе чух 'hear' крия крие крих 'hide' bijo biti бия бие бих 'beat' пен пен пях 'sing' умея умее умях 'be able' умея умее живях 'live' крия крие крих 'hide' бия бие бих 'beat' пен пях 'sing' умея умее умях 'be able' отher меля меле млях 'grind' went of the state of th		pišo	рьѕаті	II	пиша	пише	писах	'write'
Ile In shan shae shan shan shan shan shan shan shan shan				11				,
znajo znati tujo čuti kryjo kryti bijo biti pojo pěti umějo uměti delajo delati went darovati sějo sějati IV moljo moliti turpljo terpěti ležo ležati vmamb imati sejo sepati imamb imati sejo sepati imamb imati sejo znati If uýя чу́е чу́х 'hear' кри́я кри́е кри́х 'hide' би́я би́е би́х 'beat' пе́в пе́е пя́х 'sing' уме́в уме́е умя́х 'be able' ме́ля ме́ле мля́х 'grind' went delajo delati viя чу́е чу́х 'hear' кри́я кри́е кри́х 'beat' пе́в пе́е пя́х 'sing' уме́в уме́е умя́х 'be able' ответ пе́е пя́х 'sing' уме́в уме́е умя́х 'be able' ответ пе́е пя́х 'sing' уме́в уме́е умя́х 'be able' ответ пе́е пя́х 'sow' III ме́ля ме́ле мля́х 'grind' 'beg' хо́дя хо́ди хо́ди́х 'walk' III търпя́ търпи́ търпя́х 'tolerate' лежа́ лежи́ лежа́х 'recline' ответ спя́ спи́ спа́х 'sleep' III ймам йма йма ймах 'have' дя́лам дя́ла дя́лах/дела́х 'carve'					- мра	мре	мрях	'die'
čujo čuti чýя чýе чýх 'hear' kryjo kryti крия крие крих 'hide' bijo biti бия бие бих 'beat' pojo pěti пен пян пен пян 'sing' yмен умен умях 'be able' yмен умен умях 'be able' фан живен живен живнях 'live' cèr сях (сёях) 'sow' other меля меля меля меля фан meljo meljo meljo меля меля меля фан other меля меля меля фан фан II моля моли моли молих фан пред стан пред стан тран фан фан IV пред стан пред стан тран фан фан фан пред стан пред стан тран фан фан фан фан фан		znajo	znati	1	зная	знае	знаях	'know'
kryjo kryti bijo biti pojo pěti umějo uměti dělajo dělati weljo mlěti darujo sějo sějati IV moljo moliti xoždo xoditi tьгріјо tьгрěti ležo ležati v sepljo sъраti imamь imati kpůs kpůe kpůx 'hide' beat' nés née náx 'sing' ymés ymée ymáx 'be able' wubás живе́е живя́х 'live' cés cée cях (се́ях) 'sow' weljo mlěti darujo darovati sějo sějati IIa моля моли молих 'beg' ходи ходих 'walk' IIb търпя́ търпи́ търпя́х 'tolerate' лежа́ лежа́ лежа́х 'recline' other спя́ спи́ спа́х 'sleep' III ймам йма йма ймах 'have' дя́лам дя́ла дя́лах/дела́х 'carve'		žuio.	X4:	/ "				'hear'
bijo biti бия бие бих 'beat' rising' uměti uměti uměti ymėя умее умях 'be able' ymės умее умях 'be able' wubės живее живях 'live' ces ces ces cях (сеях) 'sow' other меля меле млях 'grind' lia моля моли молих 'beg' хойдо ходіті lia моля моли молих 'beg' хойдо ходіті ležo ležati търпя търпя търпя търпях 'tolerate' лежа лежа лежа 'recline' other спя спи спах 'sleep' lill имам има имах 'have' дялам дялам дяла дялах/делах 'carve'					• .	• .	* .	
роjo pěti uměti dělajo uměti dělajo dělati ymés ymés ymée ymáx 'sing' 'be able' ymés ymée ymáx 'live' cès cès cès càx (cèях) 'sow' meljo mlěti darujo darovati sějo sějati lla móля моли молих 'grind' IV moljo moliti хоždo xoditi llb търпя ходи ходих 'walk' llb търпя търпя търпя търпя търпя търпя търпя търпя търпя 'tolerate' лежа лежа лежа 'recline' other cnя спи спах 'sleep' V sърljo sъраti limamь imati ймам йма ймах 'have' дялам дялам дяла дялах/делах 'carve'				1				
umějo uměti ymés ymée ymях 'be able' dělajo dělati живе́я живе́е живя́х 'live' cėя cėe ся́х (се́ях) 'sow' meljo mlěti darujo darovati sějo sějati — IIa мо́ля мо́ли мо́ли́х 'beg' хо́до хо́діті ко́дя хо́ди хо́ди́х 'walk' IIb търпя́ търпи́ търпя́х 'tolerate' лежа́ лежа́ лежа́ лежа́ 'recline' other cnя́ спи́ спа́х 'sleep' V sърljo sъраті imamь imati ймам йма ймах 'have' дя́лам дя́лам дя́лам дя́лах/дела́х 'carve'				1				
dělajo dělati живе́я живе́е живя́х 'live' céя се́е ся́х (се́ях) 'sow' other meljo mlěti darujo darovati sějo sějati IIa monя мо́ли мо́лих 'beg' хо́хо́о хо́оіті ко́хо́дя хо́ди хо́дих 'walk' IIb търпя́ търпя́ търпя́х 'tolerate' лежа́ лежа́ лежах 'recline' other спя́ спи́ спа́х 'sleep' III ймам йма ймах 'have' дя́лам дя́ла дя́лах/дела́х 'carve'			•	1				
meljo mlěti darujo darovati sějo sějati IV moljo moliti xoždo xoditi ttrpljo trpěti ležo ležati v spljo spati imamb imati meljo mlěti mėля ме́ле мля́х 'grind' ме́ля мо́ли мо́лих 'beg' хо́дя хо́ди хо́дих 'walk' III търпя́ търпи́ търпя́х 'tolerate' лежа́ лежи́ лежа́х 'recline' other спя́ спи́ спа́х 'sleep' III ймам йма йма ймах 'have' дя́лам дя́ла дя́лах/дела́х 'carve'				1.	• .			
meljo mlěti darujo darovati sėjo sėjati — III моля моли молих 'beg' хоždo хоditi IIb търпя търпя търпях 'tolerate' лежа лежи лежах 'recline' other спя спи спах 'sleep' III ймам йма йма дялам дялам дялам дялам дялам дялам дялам дялам стату 'carve'		delajo	delati					
meljo mlěti darujo darovati sėjo sėjati— IV moljo moliti моля моли молих 'beg' хойдо ходіті tьгріјо tьгрёті ležo ležati v sъріјо sъраті imamь imati mėля меля моли молих 'beg' ходи ходих 'walk' III търпя търпя търпях 'tolerate' лежа лежа лежа 'recline' other спя спи спах 'sleep' III ймам йма йма ймах 'have' дялам дяла дялах/делах 'carve'			1	,	- сея	cee	сях (сеях)	SOW
darujo darovati sėjo sėjati— IV moljo moliti хоždo xoditi tьгрljo tьгрёti ležo ležati v sърljo sъраti imamь imati darujo darovati sėjo monjo moliti ходя моли молих 'beg' valk' подя ходи ходих 'walk' пърпя търпя търпях 'tolerate' recline' отнег спя спи спах 'sleep' III ймам йма йма ймах 'have' дялам дяла дялах/делах 'carve'		melio	mlšti	Other	vo na	No.	Mady	'arind'
sějo sějati—/ moljo moliti моля моли молих 'beg' ходо ходіti tьгріjo tьгрěti ležo ležati V sърljo sъраti imamь imati Ila моля моли молих 'beg' walk' Търпя търпи търпях 'tolerate' лежа лежа лежах 'recline' other спя спи спах 'sleep' III ймам йма ймах 'have' дялам дяла дялах/делах 'carve'					мсля	MEJIE	ARIUM	grind
IV moljo moliti хо́до хо́діті tьrpljo tьгрёті ležo ležati v sърljo sъраті imamь imati moljo moliti мо́ля мо́ли мо́ли́х 'beg' valk' търпя́ търпи́ търпя́х 'tolerate' recline' other спя́ спи́ спа́х 'sleep' III ймам йма ймах 'have' дя́лам дя́ла дя́лах/дела́х 'carve'				1				
moljo moliti мо́ля мо́ли мо́ли́х 'beg' хо́дя хо́ди хо́ди́х 'walk' Ilb търпя́ търпи́ търпя́х 'tolerate' лежа́ лежа́ лежа́х 'recline' V imamь imati ймам йма ймах 'have' дя́лам дя́лам дя́лах/дела́х 'carve'	IV	sejų	scjati —	1110				
xoždo xoditi xóдя хо́ди хо́ди́х 'walk' IIb tьгріjo tьгрёti ležo ležati тьрпя́ тьрпи́ тьрпя́х 'tolerate' лежа́ лежи́ лежа́х 'recline' other cпя́ спи́ спа́х 'sleep' III ймам йма ймах 'have' дя́лам дя́ла дя́лах/дела́х 'carve'		molio	moliti \	110	MÓRG	MÓRM	MORNA	'heσ'
tьгрljo tьгрěti ležo ležati търпя́ търпи́ тьрпя́х 'tolerate' лежа́ лежи́ лежа́х 'recline' other спя́ спи́ спа́х 'sleep' V imamь imati и́мам и́ма и́ма́х 'have' дя́лам дя́ла дя́лах/дела́х 'carve'				\				
tыrpljo tыrpěti търпя́ търпи́ търпя́х 'tolerate' ležo ležati лежа́ лежа́ лежа́ лежа́х 'recline' other спя́ спи́ спа́х 'sleep' III имам има има́х 'have' дя́лам дя́ла дя́лах/дела́х 'carve'		ποεαγ	xouiti	V пь	лоди	ходи	лодил	******
ležo ležati лежа лежа лежах 'recline' other спя спи спах 'sleep' III имам има имах 'have' дялам дяла дялах/делах 'carve'		tarplio	tsrněti	1	πъ ηπά	търпи	търпя́х	'tolerate'
sъpljo sъpati other cпи спих 'sleep' V imamь imati имам има имах 'have' дялам дялам дялах/делах 'carve'				k				
sъpljo sъpati V спя́ спи́ спа́х 'sleep' V III и́мам и́ма и́ма́х 'have' — дя́лам дя́лах/дела́х 'carve'				Other		nomn.	mun	
V imamь imati имам има имах 'have' — дялам дяла дялах/делах 'carve'		saplio	sъраti	11	спя	спи	спах	'sleep'
imamь imati имам има имах 'have' — дялам дяла дялах/делах 'carve'	V	- 26.16	o span	1\m	V/1	V.111	J	
Дялам дяла дялах/делах 'carve'		imamь	imati	1/	имам	има	и́ма́х	'have'
				11				
<u> </u>								'present gifts'

Table 5.11 continued

Old Church Slavonic Class 1 SG PRES INF			Modern Bulgarian Class 1 SG PRES 3 SG PRE		s 1 sg aor	
(b)			la			
•	neso	nesti	ıa	do-nesa	do-nesė	do-nėsox
	vedo	vesti		do-veda	do-vedė	do-vedox
	čьtо	čisti		četa	četė	čėtox
	ido	iti		ida	ide	idox
	reko	rešti		rekå	rečė	rėkox
		greti —	_	gina	gine	gináx
	grebo	žiti —		mina	· .	mináx
	živo	2111 —	\		mine	
	па-сьпо	na-čęti	\mathcal{M}	za-čėna po-čna	za-čėne po-čne	za-čėnax po-čnax
	тьго	mrěti —	//>	greba	grebe	grėbax
	stano	stati	//	stana	stane	stánáx
	kovo	kovati \/	' /	ková	kovė	kováx
	zovo	zъvati \V	/	zová	zovė	zováx
	bero	bьrati И	/	berå	berė	bráx
II	σειγ	John Ay		ociu	ocic	orux
	gybno	gybati 🗐 📉				
	mino	minati —				
Ш			lc			
	kažo	kazati	1	kaža	kaže	kázáx
	pišo	рьзаті	1	piša	piše	pisax
	Pisy	pour.	ld	piou	p.oc	Pioun
			حــــا ا	mrå	mrė	mrjáx
			le			jux
	znajo	znati	1.0	znája	znae	znájáx
	Zuj y	2	\If	znaja	Liide	Liiujux
	čujo	čuti	1	čuja	čue	čux
	kryjo	kryti	1	krija	krie	krix
	bijo	biti	1	bija	bie	bix
	pojo	pěti	1	pėja	pėe	pjáx
	umějo	uměti	1	umėja	umėe	umjax
	dělajo	dělati —	L.	- živėja	živėe	živjax
	-υ,γ			sėja	sėe	sjáx (séjáx)
		1	other	Seju	300	sjan (sejan)
	meljo	mlěti /	ounci	mėlja	mėle	mljáx
	darujo	darovati				
	sějo	sějati –				
IV	,,,	\	lla			
• •	moljo	moliti		mólja	móli	mòlix
	xoždo	xoditi		xòdja	xòdi	xòdix
	nozuy		IIb	nouju	AOG!	AOGIA
	tьrpljo	tьrpěti	•••	tărpjá	tărpi	tărpjax
	ležo	ležati		leža	leži	ležax
	-529		other			IVEMA
	sърljo	sъраti		spjå	spi	spåx
v	aphil	o span	 	opju	op.	opun
•	ітать	imati	I/***	imam	ima	imáx
	uiIID		L	djalam	djala	djálax/deláx
			L	darůvam	darůva	darůváx
						TUA

infinitive and supine, loss of present active, present passive and non-resultative past active participles, simplification of the simple imperative, generalization of a single aorist formation, restructuring of imperfect forms on the present stem, elimination of the results of dental and labial assimilations to *j (see section 2.2), and changes in individual suffixes. The third conjugation, a Bulgarian innovation, was accomplished by extending the unproductive Proto-Slavonic athematic type Old Church Slavonic 'have' PRS 1 SG imamb to many first-conjugation verbs with present theme -je-(table 5.11). This conjugation, by far the largest, is productive for derivation of secondary imperfectives and the assimilation of borrowed verbs. Otherwise, vestiges of athematic forms are limited: first person singulars pam/dam '(I) give (PRFV)' (otherwise /dad-/: дадеш/dadeš '(you-sG) give') and ¬m/jam '(I) eat (IMPFV)' (otherwise /jad-/: ¬m/jadeš); imperatives ¬m/jaž(te) 'eat (2 SG (PL))', ¬m/m/(Te)/viž(te) 'see (2 SG (PL))'.

The following paradigms illustrate the first subtype of each conjugation; those of other subtypes follow from the examples in table 5.11.

Conjugation Ia 'read'

	Present	Imperfect	Imperative	Aorist
1 sg	чета́/četа́	четях/četjax		чėтох/čėtox
2 sg	четėш/četėš	четėше/četėše	чети́/četi	че́те/čе́tе
3 sg	четė/četė	четėше/četėše		чėте/čėte
1 PL	четėм/četėm	четяٰхме/		четохме/
		četjaхме		čėtoxme
2 PL	чете́те/četéte	четяхте/četjaxte	четėте/četėte	чėтохте/čėtoxte
3 PL	четат/četat	четя́ха/četjáха		чėтоха/čėtoxa

Stem-final /t/ or /d/ is lost before /l/ of the aorist participle: M SG чел/čel. Otherwise, the consonant is retained with vowel-zero alternation: 'bake' M SG пекъл/река – F SG пекла/река. Retraction of stress to the root is regular in aorist forms.

Conjugation IIa 'go, walk'

1 sg	Present xòдя/xòdja	<i>Imperfect</i> xòдex/xòdex	Imperative	<i>Aorist</i> xòдих/xòdix
2 sg	xòдиш/xòdiš	хо́деше/xódeše	ходи/xodi	xòди/xòdi
3 sg	xòди/xòdi	xóдеше/xódeše		xòди/xòdi
1 PL	хо́дим/xódim	xо̀дехме/ xòdexme		хо́ди́хме/ xódixme
2 PL	хо́дите/xódite	хо́дехте/хо́dexte	ходете/xodete	хо́ди́хте/xódixte
3 PL	xóдят/xódjat	хо́деха/хо́dexa		xóди́ха/xódixa

This is the only conjugational type that regularly contrasts two stress patterns; compare 'pay' PRS 1 SG платя/platja, AOR 1 SG платих/platix.

Conjugation III 'have'

	Present	Imperfect	Imperative	Aorist
1 sg	и́мам/imam	и́мах/imax		и́ма́х/imáх
2 sg	и́маш/imaš	имаше/imaše	и́май/imaj	и́ма́/imá
3 sg	и́ма/ima	имаше/imaše		и́ма́/imá
1 PL	и́маме/imame	и́махме/imaxme		и́ма́хме/imáхme
2 PL	имате/imate	имахте/imaxte	имайте/imajte	имахте/imaxte
3 PL	и́мат/imat	и́маха/imaxa		и́ма́ха/imáха

The negated form of 'have' is suppletive: 1 SG PRS нямам/пјатат, 1 SG AOR нямах/пјатах and немах/петах.

The optional shift of stress from the root in the present to the suffix vowel in the aorist occurs in other aorist forms as well: M SG AOR PART хо́дил/хо́dil, имал/imal. Reflecting variation in usage, reference works do not agree on which stems allow shifted forms. In any case, verbs of foreign origin are least likely to shift: 'telephone' 1 SG PRS телефонирам/telefoniram and 1 SG AOR телефонирах/telefonirax.

The conjugational pattern of 'be' is suppletive:

	Present	Imperfect/aorist	Imperative
1 sg	съм/săm	бя́х/bjáх	-
2 sg	си/si	бė́(ше́)/bė́(še)	бъди/bădi
3 sg	e/e	бė́(ше́)/bė́(šé)	
1 PL	сме/sme	бя́хме́/bjaٰxmé	
2 PL	сте/ste	бя́хте/bjȧ́хte	бъдете/bădete
3 PL	ca/sa	бя́ха/bjȧ́ха	

Imperfect/aorist participle: м sG бил/bil, F sG била/bilå, N sG било/bilò, PL били/bili

Present forms are clitics. Alternative present forms, based on the stem of the imperative, 1 sG бъда/bada, 2 sG бъдеш/bades, often replace regular present forms in future constructions and infinitival da-constructions.

Preserving a Proto-Slavonic anomaly, the aorist past participle of /id-/ (-jd-) 'go' is suppletive: м sg -шъл/-šāl and -шел/-šel, F sg -шла/-šla, N sg -шло/-šlo, PL -шли/-šli; for example 'arrive' 1 sg prs дойда/dojda, м sg аоr ракт дошъл/došål.

3.3 Derivational morphology

3.3.1 Major patterns of noun derivation

Noun derivation is suffixal or compound. Suffixal formations are emotive or non-emotive. Productive emotive formations are (1) augmentative and/or pejorative: женище/ženište 'large woman' (< жени/ženia 'woman'), дебелан/debelan 'obese person' (< дебел/debel 'fat (M SG ADJECTIVE)'); or (2) diminutive and/or endearing: столче/stolče 'stool' (< стол/stol 'chair'), бебенце/bebence '(dear) baby' (< бебе/bebe 'baby').

Typical productive non-emotive formations are:

- /-ak-/: always stressed, masculine personal nouns: προςτάκ/prosták 1 'simpleton' (< прост/prost 'simple (M SG)').
- /-ac-/: masculine personal agent nouns from verbal bases: водач/ 2 vodáč 'leader' (< 1 SG PRS водя/vodja '(I) lead').
- /-(e)c-/: masculine personal nouns: лъжец/lažec 'liar' (< 1 SG PRS 3 лъжа́/lăžá '(I) lie').
- /-ic-/: feminine nouns, often counterparts to masculine nouns in 4 /-(e)c-/: вдовица/vdovica 'widower' (вдовец/vdovec 'widower').
- /-k-/: diverse feminine nouns, including counterparts to masculine 5 personal nouns: водачка/vodačka 'leader (F)', простачка/ prostačka 'simpleton (F)'.
- /-ost-/: abstract de-adjectival feminine nouns: вярност/vjarnost 6 'fidelity' (« верен/veren 'faithful, true (M SG)').

There are two types of compounding: (1) with 'linking' vowel /-o-/ (/-e-/): хлебопекар/xlebopekar 'bread-baker' < хляб/xljab 'bread' + neк-/pek- 'bake'; (2) simple concatenation: кандидат-член/kandidatčlen 'non-voting member' ('candidate' + 'member'.

Bulgarian adjectives are regularly used as nouns: HOBOTO/novoto 'the new one/thing (N)'. Several forms of this origin are used only in this way: 'insect' N SG насекомо/nasekomo - PL насекоми/nasekomi (as adjective).

3.3.2 Major patterns of adjective derivation

Adjectives are formed by suffixation and compounding. Productive suffixal formations include:

- 1 /-(e)n-/: qualitative and relational adjectives: 'successful' M SG успешен/uspešen, F SG успешна/uspešna (< успех/uspex 'success').
- /-in-/: possessive relational adjectives from animate feminine and 2 masculine nouns with singulars in /-a/: сестрин/sestrin 'sister's (м SG)' (cecтpa/sestra 'sister').
- /-ov-/: possessive relational adjectives from masculine personal 3 nouns: братов/bratov 'brother's (M SG)' (< брат/brat 'brother').

Compound formations use (1) linking with /-o-/ (/e/), злочест/ zločest 'unfortunate (M SG)' < зъл/zal 'evil (M SG)' + чест/čest 'fate', or (2) simple concatenation, полуграмотен/polugramoten 'semi-literate (M SG)' (полу-/polu- 'semi' + грамотен/gramoten 'literate (M SG)'.

See section 3.1.4 for comparative and superlative adjectives. Two other stressed prefixes affect the degree of adjectives: (1) Bb3-/vaz- attenuates the adjective: въззелен/vazzelen 'greenish (M SG)'; (2) пре-/preheightens it: предобър/predobar 'too good, overly good (M SG)'.

Major patterns of verb derivation

A small number of Bulgarian non-prefixed imperfective stems are non-derived, such as $\pi \in \text{--}/\text{pek-'bake'}$. Most are formed with a limited number of verbalizing suffixes, which appear in the present and/or agrist stems; for example:

- present stem in /-Ø/ ~ aorist stem /-a-/: 'write' 1 sg PRS пиша/piša, 1 1 SG AOR писах/pisax (< пис-/pis- 'write');
- present stem /-ėj-/ ~ aorist stem /-å-/: 'live' 1 SG PRS живея/živėja, 2
- 1 SG AOR живя́х/živjáх (< жи́в/živ 'alive (M SG)'); present stem in /-Ø/ ~ aorist stem /-i-/: 'go, walk' 1 SG PRS хо́дя/хо́діа, 1 SG AOR хо́дих/хо́діх (< хо́д/хо́д 'course, movement'). 3

In the process of deriving perfective stems from non-prefixed imperfectives the meaning of the verbal base may be augmented (1) by the meaning of the perfectivizing prefix or (2) by the semelfactive meaning of the perfectivizing suffix /-n-/. For examples see section 3.2.1.

4 **Syntax**

Element order in declarative sentences

The unmarked order of main constituents in simple declarative sentences is Subject-Verb-Object: Иван отвори вратата/Ivan otvori vratata 'Ivan opened the door'. Other orders change the logical stress or other stylistic nuances: вратата отвори Иван/vratata otvori Ivan 'Ivan opened the door' (Rudin 1986: 14ff.; Пенчев/Репсе 1984: 89ff.).

Adverbial modifiers in verb phrases normally come at the end of the sentence: Иван ще дойде тук рано утре/Ivan šte dojde tuk rano utre 'Ivan will come here early tomorrow'.

Subjectless sentences are common. Some result from the omission of subject pronouns (see section 4.7). Others - with neuter third person subject pronouns (see section 4.7). Others – with neuter third person singular verbs – seem to be genuinely **impersonal**; these include (1) verbs denoting natural phenomena (вали/vali 'precipitates'), (2) the copula 'be' plus adjectives (горещо e/gorešto e '(it) is hot'), (3) reflexivized verbs in constructions denoting desire/lack of desire, permission or prohibition (see section 3.2.1), (4) modal verbs (such as бива/biva 'ought', трябва/ trjabva 'needs, must'), (5) certain copular verbs (such as струва се/struva se 'seems', изглежда/izgležda 'appears').

Bulgarian has many clitics – phonologically unstressed words – whose positions with respect to other constituents are fixed. Proclitics include (1) prepositions (из нашата страна/iz nasata strana 'throughout our country'), (2) the negative particle нe/ne 'not' (see section 4.6), (3) the future auxiliary ще/šte (section 3.2.1), (4) the infinitival complementizer да/da (искам да пиша/iskam da piša '(I) want to write'), (5) the conjunctions и/i 'and', но/no 'but', a/a 'and/but' and the complementizer че/če 'that' (и мене/i mene 'me too' (= 'and me')).

Enclitics include (1) the definite article (see section 3.1.1), (2) the interrogative particle πu/li (section 4.2), (3) 'short' dative personal pronouns denoting possession (section 4.9).

Two groups of verbal clitics – (1) non-emphatic dative and accusative personal pronouns (as indirect and direct objects, respectively) and (2) present forms of 'be' – are variable: when the verb phrase is the initial constituent of the sentence, they are enclitics on its first stressed constituent; otherwise they are proclitics. In addition, dative pronominal clitics precede accusatives; third person singular 'be' follows pronominal clitics while other persons and numbers precede them:

Дал си му ги./Dal si mu gi. gave-м SG be-2 SG-PRS him-DAT them-ACC '(You-SG) have given them to him.'

Дала му ги e./Dala mu gi e. gave-F SG him-DAT them-ACC be-3 SG-PRS '(She) has given them to him.' Tи си му ги дал./Ti si mu gi dal. you-sg be-2 sg-prs him-dat them-ACC gave-м sg 'You-sg have given them to him.'

Тя́ му ги е дала./Тjá mu gi e dala. she him-DAT them-ACC be-3 SG-PRS gave-F SG

'She has given them to him.'

For details on the interaction among various rules of clitic placement see Hauge (1976) and Ewen (1979).

The order of constituents within simple noun phrases is:

demonstrative + possessive + quantifier + adjectival + noun + prepositional pronoun pronoun phrase(s) phrase

The minimal noun phrase is a single noun; the maximal noun phrase contains all of the above elements: тези мои две много скъпи нови книги от Германия/tezi moi dve mnogo skapi novi knigi ot Germanija these my two very expensive new books from Germany.

Quantifiers include cardinal numerals and other quantifying words, such as малко/malko 'few, little' and много/mnogo 'many, much' (see section 4.10). Only cardinal numerals co-occur with pronouns; *тези мои много книги/tezi moi mnogo knigi 'these my many books' is impossible.

The definite article and demonstrative pronouns do not co-occur: те́зи книги/tézi knigi 'these books' or книгите/knigite 'the books'.

Adjectival phrases contain any number of adjectives. Adjectives are modified by adverbs of quantity or degree, which regularly precede them: много интересен човек/mnogo interesen čovek 'a very interesting

person'. Adverbs of this type also modify predicate adverbials (above): много хубаво/mnogo xubavo 'very nicely'.

Adjectives and other inflected noun-phrase attributes may follow their heads. Very rhetorical, this usage is common in poetry, both literary and folk: векове цели/разум и съвест с нея се борят .../vekove celi/razum i savest s neja se borjat ... 'entire centuries/reason and conscience struggle with it ...' (= 'centuries entire'; Xristo Botev, cited Граматика/Gramatika 3: 286).

4.2 Non-declarative sentence types

There are two types of interrogative sentences.

1 Yes-no questions use several question-markers; much less frequently, they are formally identical to declarative sentences but carry a final rising intonation (rather than a falling, declarative intonation).

Neutral yes-no questions use the particle $\pi u/li$. The particle follows the first stressed constituent of the verb phrase, if the scope of interrogation is the entire sentence; otherwise it follows the interrogated constituent. In either case, the constituent to which the particle is attached begins the question:

Искат ли ракия?/İskat li rakija? 'Do (they) want brandy?' Ракия ли искат?/Rakija li iskat? 'Is it brandy (they) want?'

Rarely, the questioned constituent is not initial: можеш да плуваш ли?/ možeš da pluvaš li? and да плуваш ли можеш?/da pluvaš li možeš? 'can (you-SG) swim?' (Граматика/Gramatika 3: 53).

In neutral questions an overt subject may follow or precede the interrogated verb phrase. The latter construction, which topicalizes the subject, is marked by a sharp intonational rise over the subject followed by a slight pause before the predicate: ще дойде ли Иван?/šte dojde li Ivan? ~ Иван | ще дойде ли?/Ivan | šte dojde li? 'will Ivan come?'.

Дали/dali introduces a strongly rhetorical question or one which the speaker regards as difficult to answer in some sense: дали ще ни приемат?/dali šte ni priemat? 'will (they or won't they) accept us?' (Граматика/Gramatika 3: 54).

Нали/nali introduces a question anticipating a positive answer: нали сте болен?/nali ste bolen? '(you-м SG POLITE) are sick(, aren't you)?'. It is also the universal tag question after declarative sentences: той е болен, нали?/toj e bolen, nali? 'he's sick, isn't he?'.

Нима/nima forms questions anticipating negative answers, often with a sense of surprise, shock or doubt: нима той ме е видял?/nima toj me e vidjal? 'did he see me?' (= could he possibly have seen me?).

A positive simple question is answered да/da 'yes' or не/ne 'no' to express the hearer's assertion of the truth or falseness of the corresponding statement:

Иван, изпрати ли писмото?/ Ivan, izprati li pismoto?

'Did Ivan send the letter?'

True: Да, (изпрати го)./Da, (izprati go).

'Yes, ((he) sent it).'

False: He, (не го изпрати)./Ne, (ne go izprati).

'No, ((he) didn't send it).'

Simple negative questions, formed with the negative particle He/ne, are answered on the basis of the hearer's assertion of the truth or falseness of the corresponding positive statement:

Не изпрати ли Иван писмото?/Ne izprati li Ivan pismoto?

'Didn't Ivan send the letter?'

He did not send it: He, (не го изпрати)./Ne, (ne go izprati).

'No, ((he) didn't send it).'

He did send it: Да, изпрати го./Da, izprati go.

'Yes, (he) sent it.'

As an answer to a negative question, да/da alone is unacceptable. See Лакова/Lakova (1978; examples hers).

2 Constituent questions are formed with interrogative pronouns (see section 3.1.3) and interrogative adverbs such as как/kak 'how', къде/kade 'where', кога/koga 'when' and защо/zašto 'why'. Such wh words generally begin questions: какво виждаш?/kakvo viždaš? 'what do (you-SG) see?'. Prepositions always precede the wh words they govern: с кого работиш?/s kogo rabotiš? 'with whom are (you-SG) working?'. Multiple wh words normally occur in the beginning of the sentence: кой какво кога казва?/koj kakvo koga kazva? 'who says what when?' (= 'who what when says'). Questions with non-initial wh words are interpreted as echo questions: ти си видял кого?/ti si vidjal kogo? 'you-SG saw whom?' (Rudin 1986: 82).

WH words and ли/li co-occur in two ways. (1) ли/li after an initial WH word makes a strongly rhetorical question: и каква ли сладост намират в него?/i kakva li sladost namirat v nego? 'and what sweetness do (they) find in it?'. (2) Sentence-final ли/li makes an echo question: какво ще ям ли?/kakvo šte jam li? 'what will (I) eat?', as an echo to какво ще ядещ?/kakvo šte jadeš? 'what will (you-sg) eat?' (Граматика/ Gramatika 3: 61).

Simple questions with дали/dali or, less commonly, ли/li and questions with WH words may be embedded in diverse syntactic positions:

He зная дали е там./Ne znaja dali e tam. ~ He зная там ли e./Ne znaja tam li e. '(I) don't know whether (he/she/it) is there.'

(Дали/dali is more common in indirect questions than ли/li; Rudin 1986: 63.)

Питаха ме, кой те е търсил./Pitaxa me, koj te e tarsil. '(They) asked me who was looking for you-sg.'

Commands are expressed with simple or compound imperatives. Simple imperatives are limited to second person singular and plural (see section 3.2). Negative simple imperatives use (1) the negated simple imperative or (2) недей(те) да/nedėj(te) da + second person present tense: не пиши!/ ne piši! ~ недей да пишеш!/nedėj da pišeš! 'don't write (you-SG)!'. Rarely, недей (те)/nedėj(te) is followed by the vestigial infinitive (section 3.2): недей писа!/nedėj pisa! Only simple imperfective imperatives can be negated. Positive imperatives are possible for both imperfective and perfective verbs; imperfectives carry a sense of urgency lacking in perfectives.

Compound imperatives use да/da or нека (да)/neka (da) + present tense:

Нека аз (да) пиша!/Neka az (da) piša!
'Let me write!'
Да знаеш какво става!/Da znaeš kakvo stava!

'(You-SG) should/ought to know what's happening!/May (you-SG) know what's happening!'

Second-person formations with да/da are more categorical than simple imperatives. Forms with нека/neka carry an element of exhortation; they are uncommon in the second person (Граматика/Gramatika 3: 69ff.).

Present- and future-tense forms spoken with the intonational pattern of simple imperatives commonly serve as imperatives: ще пишеш и ти!/šte pišeš i ti! 'and you-SG too will write!'.

4.3 Copular sentences

Copular sentences use 'be' or one of a small number of other verbs, notably бивам/bivam '(I) happen (to be)', оставам/ostavam '(I) remain', ставам/stavam '(I) become'. Predicate adjectives, including participles, agree in number and gender with the subject: Мама изглежда болна/ Mama izgležda bolna 'Mother looks sick-F SG'. With second person plural personal pronouns referring to singular addressees for politeness (see section 3.2), participles in compound verbal forms and predicative adjectives are singular and masculine or feminine as appropriate; simple verbal forms are plural: вие сте била много добра/vie ste bila mnogo dobra

'you (= F SG) have been very kind' (= 'you-PL are-2 PL be-F SG-PAST PART very kind-F SG'). In formal writing and speech adjectives and participles in these constructions are often plural.

Perceived qualities or states are commonly expressed by impersonal sentences using neuter adjectives, 'be' and optional prepositional phrases with Ha/na or clitic dative personal pronouns:

Студено ми e./Studeno mi e.
'I feel cold.' (= cold-N SG me-DAT is')
На Иван му е много скучно./Na Ivan mu e mnogo skučno.
'Ivan is very bored.' (= 'to Ivan him-DAT is very boring-N SG'; for duplicated pronoun see 4.7)

A similar construction with a small number of nouns, like crpax/strax 'fear', cpam/sram 'shame', uses an accusative pronoun: cpam го e/sram go e 'he is ashamed' (= 'shame him-ACC is') (Граматика/Gramatika 3: 95).

4.4 Coordination and comitativity

The coordination of all types of phrasal constituents is possible with the conjunction и/i 'and' preceding the last coordinated constituent: купих си риза, чадър и шапка/киріх si riza, čadar i šapka '(I) bought myself a shirt, an umbrella and a hat'. Repetition of the conjunction before all conjoined elements stresses their parallelism: и Петър и Иван отидоха на мач/і Petar i Ivan otidoxa na mač 'both Peter and Ivan went to the game'.

Complete sentences are conjoined with и/i: дъжд вали и вятър духа/dažd vali i vjatar duxa 'rain is falling and wind blowing'. Here repetition or deletion of the coordinating conjunction adds a stylized, literary flavour:

И слънцето се въртеше, и времето летеше и изтичаше като стадо, като река, като облак./І slanceto se vartese, i vremeto letese i iztičase kato stado, kato reka, kato oblak.

'And the sun was revolving, and time was flying and running out like a flock, like a river, like a cloud.' (A. Dončev, cited in Граматика/Gramatika 3: 294)

Coordinated subjects generally show plural verbal concord (examples above). Singular agreement is also possible, for example (1) with paired subjects construed as a unit: в колата екна смях и кикот/v kolata ekna smjax i kikot 'in the car (there) resounded-3 SG laughter and giggling'; or (2) with coordinated series: ние не ще кажем нищо друго освен това че всеки човек и всеки народ тегли от ума си/піе пе šte kažem ništo drugo osven tova če vseki čovek i vseki narod tegli ot uma si 'we shall say nothing except that every person and every nation is responsible for

itself'. For additional details on singular agreement with plural subjects see Граматика/Gramatika 3: 146-54 (preceding examples there).

Comitative constructions use a prepositional phrase with c/s 'with': майка с дете чакаше на опашката/majka s dete cakase na opaskata 'a mother (together) with a child waited (SG) in the line'. Pronominal heads of comitative constructions, regardless of their actual number, are plural, as are related verbs: ние с баща ми отидохме на кино/nie s basta mi otidoxme na kino 'I went to the movies with my father/my father and I went to the movies' (= 'we with father me-DAT ...'). Comitative constructions are possible as direct and indirect objects:

Арестуваха ги с баща му./Arestuvaxa gi s bašta mu.

'(They) arrested him and his father.' (= '... them-ACC with father him-DAT') На нас със жена ми ни изпратиха подарък./Na nas sas žena mi ni izpratixa podarak.

'(They) sent me and my wife a gift.' (= 'to us-ACC with wife me-DAT us-DAT ...')

Other types of coordination utilize other conjunctions: или/ili 'or', или ... или/ili ... ili 'either ... or', a/a 'and, but' (contrasting opposition), но/ no 'but, however' (strong contrast, contradiction), обаче/obače 'however' (stronger, more literary form of но/no):

Аз имам чадър, а тя няма./Az imam čadar, a tja njama.

'I have an umbrella, but she doesn't.'

Пиесата има голям успех, но въпреки това падна./Piesata ima goljam uspex, no vapreki tova padna.

'The play had great success, but nevertheless closed (literally 'fell').'

4.5 Subordination

Headed relative clauses, postposed to noun phrases, are marked by initial relative pronouns (see section 3.1.3), preceded by prepositions governing them. The number and gender of relative pronouns agree with their antecedents; case is determined by the function of the pronouns in the relative clauses: изпратих пакет на студента, с когото се запознахме миналата година в София/izpratix paket na studenta, s kogoto se zapoznaxme minalata godina v Sofija '(I) sent a package to the student, with whom I got acquainted (= with whom-ACC self-ACC acquainted-1 PL) last year in Sofia'. Relative clauses using relativizers formed from interrogative adverbs modify prepositional and adverbial phrases: намерих книгата там, където я бях оставил/патетх knigata tam, kadeto ja bjax ostavil '(I) found the book there, where (I) had left it'.

Relative pronouns with noun phrase antecedents have alternative formations with the invariant relative pronoun дето/deto 'that'. Most frequent in dialects, this construction is colloquial in literary Bulgarian: това е книгата дето (~ която) беше на масата/tova e knigata deto (~ kojato) beše na masata 'that is the book that (which) was on the table'.

If дето/deto is not the subject of its clause, the clause may contain a clitic personal pronoun 'echoing' the grammatical categories of the antecedent; in relativization from a prepositional phrase a full, stressed personal pronoun is necessary as the object of the preposition:

Това е книгата дето (я) купих./Tova e knigata deto (ја) kupix.

'This is the book that (it) (I) bought.'

Това е книгата дето говорежме за нея./Tova e knigata deto govorexme za

'This is the book that (we) spoke about (it).'

See Rudin (1986: 129-30) for discussion; the examples are hers.

Free (headless) relatives are possible:

Вземи каквото искаш!/Vzemi kakvoto iskaš!

'(You-SG) take whatever (you-SG) want!'

Който не внимава, той прави грешки./Kojto ne vnimava, toj pravi greški.

'Whoever doesn't pay attention, makes mistakes.'

In multiple free flatives all WH words are clause initial: който където е свикнал, там си живее/kojto kadeto e sviknal, tam si živee 'everyone lives where he's accustomed' (= 'whoever wherever is becameaccustomed-M SG, there self-DAT lives'; Rudin 1986: 167).

The complementizer 4e/če 'that' introduces subordinate clauses as complements of verbs of saying or believing, as complements of nouns and as sentential subjects:

Мисля че скоро ще завали дъжд./Mislja če skoro šte zavali dažd.

'(I) think that soon (it) will rain.'

Имам чувство че ще дойдат./İmam čuvstvo če šte dojdat.

'(I) have a feeling that (they) will come.'

Сигурно е че няма да дойдат./Sigurno e če njama da dojdat.

'(It) is certain that (they) won't come.'

In place of an infinitive, Bulgarian uses a subordinate clause consisting of да/da plus a present-tense verbal form which agrees in person and number with its implicit subject. Infinitive-like 'da-clauses' are used as complements of many modal verbs (for example, трябва/trjabva '(it) is necessary', искам/iskam '(I) want', мога/moga '(I) can'), as complements of verbs of motion or being, as nominal complements, as sentential subjects and in indefinite constructions with 'have' and 'have not':

Йскам да отида на мач./ İskam da otida na mač.

'(I) want-1 sG to go-1 sG to a game.'

Дошли сме да слушаме музика./Došli sme da slušame muzika.

'(We) have come to listen-1 PL to music.'

Нямам намерение да се напрягам./Njamam namerenie da se naprjagam.

'(I) don't have (the) intention to exert-1 sG myself.'

Ймам с кого да говоря./İmam s kogo da govorja.

'(I) have someone to talk to.' (= 'have-1 sG with whom-ACC to talk-1 sG')

The subject of a main clause may differ from that of a da-clause: дошли сме да ни разкажете за пътуването си в България/došli sme da ni razkažete za pătuvaneto si v Bălgarija '(we) have come for (you-PL) to tell us about your (own) trip to Bulgaria'. See Rudin (1986) for discussion of Bulgarian complementizers and relative clauses.

The vestigial **infinitive** (see section 3.2.1) occurs rarely in the literary language as the complement of мо́га/moʻga '(I) can' and сме́я/sme˙ja '(I) dare', and in negative imperatives (section 4.2): не мо́га писа/пе moʻga pisa '(I) can't write'. *Da*-clauses are normal here: не мо́га да пиша/пе moʻga da piša '(I) can't write'.

Diverse subordinate clauses expressing circumstances related to the actions of main clauses are formed with subordinating conjunctions.

- 1 Conditional clauses use the conjunction ako/ako 'if' (or its colloquial, poetic variant да/da). Real conditionals use indicative verbal forms in both condition and result clauses: ако дойдат, ще ги видим/ako dojdat, šte gi vidim 'if (they) come, (we) will see them'. Contrary-to-fact conditionals normally use imperfect or past perfect indicative forms in the condition clause and subjunctive or past future indicative forms in the result clause; these tense distinctions apparently entail no semantic differences: бих я поздравил (щих да я поздрави), ако бих там/bix ја pozdravil (štjax da ја роzdravja), ako bjax tam '(I) would greet/would have greeted her, if (I) were/had been there'.
- 2 Other subordinate clauses function as predicate adverbials of time, place, manner, reason and so on:

Говориш, без да знаеш какво става./Govoriš, bez da znaeš kakvo stava. '(You-sG) speak without knowing (= without to know-2 sG) what is happening.' Доколкото ми е известно, няма да дойдат./Dokolkoto mi e izvestno, njama da dojdat.

'As far as (it) is known to me, (they) will not come.'

In written and formal spoken Bulgarian participial phrases often replace relative clauses. An active participle – either present or aorist – replaces a subject relative pronoun and its verb. The participle is present tense if the action of the verb in the relative clause is contemporaneous with that of the main verb; it is aorist if the action of the verb of the relative clause precedes that of the main verb:

Познавах протестиращите хора./Poznavax protestirastite (PRS ACT PART) хога. '(I) knew the protesting people.'

Намерихме пристигналия влак./Namerixme pristignalija (AOR ACT PART) vlak. '(We) found the arrived train.'

A past passive participle replaces a transitive verb and direct object relative pronoun: намерих загубената книга/namerix zagubenata (PAST PASS PART) kniga '(I) found the lost book'.

The sense of a present passive participle (otherwise lacking) is expressed by an imperfective past passive participle: носена от слабия утринен вятър, [отровна мъгла] запълзя .../nosena (IMPFV PAST PASS PART) ot slabija utrinen vjatar, [otrovna magla] zapalzja ... 'carried by the weak morning wind, [a poisonous fog] drifted ...' (Маслов/Maslov 1981: 262).

Participial phrases regularly allow diverse predicative elements: много са желаещите да почиват тук/mnogo sa želaeštite (PRS ACT PART) da počivat tuk 'many are those wishing (literally 'wishing-the') to vacation here' (Narodna mladež, 1 December 1989). While usually preposed to their heads (see above), participial phrases may also be postposed: ... след кардиналните промени, настыпили в БКП/... sled kardinalnite promeni, nastapili (AOR ACT PART) v BKP '... after the fundamental changes, which have taken place in the BCP (= Bulgarian Communist Party)' (Narodna mladež, 1 December 1989).

An adverbial subordinate clause may be replaced by a **gerundive** phrase if (1) the subjects of both clauses are the same and (2) the action of the subordinate clause is contemporaneous with that of the main clause: четейки новия му роман, намерих .../četejki novija mu roman, namerix ... '[while] reading his new novel, (I) found ...'. Postposed active participles occasionally function as gerunds: машината теглеше леко и плавно, не усещаща товара/masinata tegleše leko i plavno, ne useštašta (PRS ACT PART) tovara 'the car drew away easily and smoothly, not feeling the load' (Маслов/Maslov 1981: 282-3).

Gerundive phrases are limited to formal writing and speech, and even there they are avoided. A study of a corpus of 15,000 words of a modern Russian novel and its Bulgarian translation found that of 107 gerunds in the Russian original, only fifteen (14 per cent) were translated as Bulgarian gerunds; the rest were replaced by subordinate clauses (Дончева/ Dončeva 1975).

The extraction of interrogative and relative pronouns from various types of embedded clauses has been studied by Rudin (1986). The following summarizes her major findings.

Questions cannot be formed by extraction from indirect (embedded) questions: *кого се чудиш кой е видял?/kogo se čudiš koj e vidjal? 'whom do (you-sg) wonder who saw?'. However, interrogation and extraction are possible over the complementizer че/če 'that': кого мислиш че е видял?/kogo misliš če e vidjal? 'whom do you think that he

saw?' (examples from Rudin 1986: 105f.).

Relative-clause formation can move relative pronouns an indefinite distance: книгата, която ти ми каза че Иван мисли че Петър иска да купи .../knigata kojato ti mi kaza če Ivan misli če Petar iska da kupi ... 'the book which you-sG told me that Ivan thinks that Peter wants to buy ...'. Similarly, the relativizer дето/deto 'that' is interpreted over an unbounded domain: книгата дето ти ми каза че Иван мисли че Петър иска да (я) купи .../knigata deto ti mi kaza če Ivan misli če Petar iska da (ja) kupi ... 'the book that you-sG told me that Ivan thinks that Peter wants to buy (it) ...'.

The formation of wH relative clauses is blocked in two instances:

- 1
- The determiner of a noun phrase cannot be relativized: *това е момчето (на) което трябва да намерим (неговата) майка/ *tova e momčeto (na) koeto trjabva da namerim (negovata) majka 'this is the boy (of) whom (we) must find (his) mother'. Relativization out of a noun phrase containing a head noun and a modifying clause is impossible: *това е момчето на което мисълта че (му) дадохме бонбони ядосва лекаря/*tova е momčeto na koeto misalta če (mu) dadoxme bonboni jadosva lekarja 'this is the boy to whom the thought that (we) gave him candies angers the doctor' 2 the doctor'.

Relativization with дето/deto is grammatical in both cases. See Rudin (1986: 140-3) for additional discussion (examples hers).

The complementizer че/če 'that' cannot in general be deleted, including

cases with movement across the complementizer:

Мисля че си видял човека./Mislja če si vidjal čoveka.

'(I) think that (you-sG) saw the man.'

Not *Mисля си видя́л .../*Mislja si vidjal ...

'(I) think you saw ...'

Познавам човека когото мисля че си видял./Poznavam čoveka kogoto mislja če si vidial.

'(I) know the man whom (I) think that you saw.'

Infrequent examples with deletion of the complementizer are reported in direct speech (or thought): мисля си, ти си видял човека/mislja si, ti si vidjal čoveka '(I) think (to myself): you-SG saw the man' (Граматика/Gramatika 3: 344).

4.6 **Negation**

A sentence or any of its constituents can be negated. If the scope of negation is the entire sentence, the negative particle не/ne attaches to the verb: той не работи/toj ne raboti 'he doesn't work'. Otherwise, не/ne

precedes the negated constituent, which generally begins the sentence: He на Иван дадох книгата/ne na Ivan dadox knigata 'it wasn't Ivan I gave the book to' (= 'not to Ivan gave-1 SG book-the').

Negative sentences are conjoined with ни(то)/ni(to) 'neither, nor', repeated before each conjoined constituent: нито Иван, нито Валя, нито Димчо не знаят, къде живея/nito Ivan, nito Valja, nito Dimčo ne znajat, kade živėja 'neither Ivan, nor Valja, nor Dimčo know where I live'.

Negative constituent questions (see section 4.2) use negative pronouns and other negative pro-forms (section 3.1.3); here He/ne accompanies the verb as well:

На никого не казах новината./Na nikogo ne kazax novinata. '(I) told no one the news.' (= to no-one-ACC not told-1 sG news-the') Никога не слушат./Nikoga ne slušat. '(They) never listen.'

4.7 **Anaphora and pronouns**

The principal anaphoric pronouns of Bulgarian include:

- third-person personal pronouns: той/toj 'he, it' 1
- personal possessive pronouns: HeroB/negov 'his, its' 2
- demonstrative pronouns: този/tozi 'this (M SG)' онзи/onzi 'that (M SG)' такъв/takav 'such (M SG)'

Forms in /on-/ are marked as distanced from the speaker; forms in /t-/ are unmarked.

- relative pronouns: който/kojto 'who, which, that (M SG)' 4 какъвто/kakavto 'which sort (M SG)'
- reflexive pronouns: ceoe си/sebe si '-self' 5

(See section 3.1.3 for full paradigms; for relative, reflexive and personal possessive pronouns see also sections 4.5, 4.9 and 4.10.) For example:

Speaker A: Тя пише нова граматика./Тја piše nova gramatika.

'She's writing a new grammar.' Speaker В: Тази граматика не ме интересува/Tazi gramatika ne me interesúva.

'This grammar doesn't interest me.'

or В: Това не ме интересува./Tova ne me interesuva. 'This doesn't interest me.'

Anaphoric adverbs are formed from the demonstrative root /t-/ 'this'; for example, там/tam 'there', тука/tuka 'here', тогава/togava 'then': Откриха нов магазин. Там продават всичко./Otkrixa nov magazin. Tam prodavat vsičko '(They) found a new store. There (they) sell everything.'

The omission of pronominal subjects, including first and second person, is extremely common. In a comparison of a Russian novel and its Bulgarian translation, Дончева/Dončeva (1975) found that Bulgarian omitted 58 per cent of possible personal pronominal subjects (Russian only 42 per cent). Omission is naturally facilitated by verb and adjective concord with the subject: болен e/bolen e '(he) is sick-м sg'. Often, however, omission leaves the subject ambiguous within its clause: в момента пише/v momenta piše 'at the moment (he/she/it) writes'. In such cases the larger context provides the appropriate interpretation of the intended subject.

The presence or absence of overt personal pronominal subjects may be affected by discourse factors. According to Machob/Maslov (1981: 356), the pronominal subject is omitted when topic, but retained when focus (where it generally identifies the actual agent among a number of possibilities); compare: днес отивам на кино/dnes otivam na kino 'today (I) am going to the movies' versus днес аз отивам на кино/dnes az otivam na kino 'today I (as opposed to someone else) am going to the movies'.

na kino 'today I (as opposed to someone else) am going to the movies'.

'This' and 'that' used alone as subjects of copular verbs are invariably neuter singular, and verbal concord is with the predicate: това са мо́ите но́ви дре́хи/tova sa moite novi dreٰxi 'these (= this-N SG) are my new clothes'.

The definite article marks the presupposition that the referent of the noun phrase is known to the participants in the speech or narrated event: книгата е интересна/knigata e interesna 'the book is interesting'. See Mayer (1988) for details.

'The former' and 'the latter' are expressed by the definite ordinal numerals първият/parvijat 'the first (м sg)' and вторият/vtorijat 'the second (м sg)', inflected to agree with their antecedents: първият [пример] показва ..., а вторият .../parvijat [primer] pokazva ..., а vtorijat ... 'the former [example-м sg] shows ..., while the latter ...'. Longer series use successive ordinal numerals beyond 'second'.

In colloquial Bulgarian, and to some extent in more formal styles, nouns or full, stressed personal pronouns as direct or indirect objects are often 'echoed' by appropriately inflected clitic personal pronouns (Γραματικα/ Gramatika 3: 186–8, 191). While clitic doubling is generally optional, it is subject to discourse factors: an emphasized noun phrase is apt to occur first in the sentence – the position of focus – and if the noun is the direct or indirect object the clitic pronoun is apt to appear (Rudin 1986: 139). Compare:

Neutral: Казах новините на Иван./Kazax novinite na Ivan. '(I) told the news to Ivan.'

Focus on Ivan: На Иван му казах новините./Na Ivan mu kazax novinite.

'(I) told the news to Ivan.' (= 'to Ivan him-DAT told-1 sg news-

the')

Focus on news: Новините ги казах на Иван./Novinite gi kazax na Ivan. '(I) told the news to Ivan.' (= 'news-the them-ACC told-1 SG to Ivan')

The pronoun is obligatory if the subject and direct object are ambiguous: майката я гледа детето/majkata ja gleda deteto 'the child takes care of the mother' (= 'mother-the-F SG her-ACC takes-care-of child-the-N SG') versus детето го гледа майката/deteto go gleda majkata 'the mother takes care of the child' (= 'child-the-N SG it-ACC takes-care-of mother-the-F SG').

Echoed clitics are not possible in WH-relative clauses: *човека, който Борис го видя/čoveka, kojto Boris go vidja 'the person, whom Boris saw' (= 'man-the who Boris him-ACC saw-3 sG') (Rudin 1986: 126, for discussion; example hers). On the other hand, echoed clitics with дето/deto-relatives are optional or obligatory depending on the syntactic function of the relative pronoun (see section 4.5 for examples and references).

Bulgarian grammars describe stressed personal pronouns as 'echoes' of subject nouns. Here, however, the nouns are in apposition to the personal pronouns: тя, бедната дама, започна да плаче/tja, bednata dama, započna da plača 'she, the poor lady, began to cry' (Граматика/ Gramatika 3: 123; also Пенчев/Репčev 1984: 83).

4.8 Reflexives and reciprocals

Reflexive personal pronouns (see section 3.1.3) replace non-subject personal pronouns whose antecedents are the subject of the clause in which they occur: мия се/mija se '(I) wash myself'. Stressed forms – in the case of the dative a prepositional phrase with на/па + accusative – are used for emphasis: себе си мия/sebe si mija '(I) wash myself'. In other, non-dative prepositional phrases only stressed accusative forms are possible: той мисли само за себе си/toj misli samo za sebe si 'he thinks only about himself'.

In colloquial Bulgarian first- and second-person non-reflexive forms may replace reflexives in prepositional phrases: вземи ме с те́бе!/vzemi me s tébe! ~ ... със се́бе си!/... săs sébe si! 'take me with you/yourself!'. Although considered non-standard, this usage is gaining ground. Substitution for direct or indirect objects is ungrammatical: only купи си но́ви дре́хи!/kupi si noٰvi dréхi!, not *купи ти ~ *на те́бе...!/*kupi ti ~ *na tëbe ...! 'Buy-2 sg yourself new clothes!'.

Substitution of personal for reflexive pronoun in the third person regularly changes the pronominal reference: той го мие/toj go mie 'he washes him (= someone else)'.

The reflexive pronoun always refers to the subject of the verb of the clause in which it occurs. Through the subject, even if deleted, it may refer to earlier nouns and pronouns:

Иван иска да си купи нова риза./Ivan iska da si kupi nova riza. 'Ivan wants to buy-3 sG himself a new shirt.'

L_______

Possessive expressions (see section 4.9) in predicates use reflexive personal pronouns and possessive adjectives in much the same way: aٰ3 взе́х книгата си/aٰz vzéx knigata si 'I took my (own) book'. The dative enclitic reflexive is regularly replaced by the reflexive possessive adjective for emphasis: aٰ3 взе́х сво́ята книга/aٰz vzéx svojata kniga 'I took my own book'. In the first and second person, non-reflexive possessives are possible for greater emphasis: взе́х мо́ята книга/vzéx mojata kniga 'it was my book that I took' (= 'took 1-SG my book'). Here dative non-reflexive forms are ungrammatical or questionable:

*Взех книгата ми./Vzex knigata mi.

'(I) took my book.'

?Дадох му книгата ми./Dadox mu knigata mi.

'(I) gave him my book.'

(Маслов/Maslov 1981: 302)

Reflexive possessives cannot be subjects or predicative nominatives: *нужна ми е своята стая/*nužna mi e svojata staja 'I need my own room' (= 'necessary-F SG me-DAT is own-F SG room-F SG').

With plural subjects and verbs of appropriate meanings, reciprocal constructions are formed with enclitic reflexive pronouns, dative and accusative and/or the expression един друг/edin drug 'one another'. Enclitic pronouns are customary for direct and indirect objects: те се целуват/te se celuvat 'they kiss (one another)'. Един друг/edin drug is added for emphasis: те се целуват един друг/te se celuvat edin drug 'they kiss one another'. Един друг/edin drug is the unmarked form of this expression; feminine and neuter forms are used only with subjects that are all feminine or all neuter, respectively.

In other, non-dative prepositional phrases reciprocity is expressed only by един друг/edin drug: те намират интересни неща един в друг/te namirat interesni nešta edin v drug 'they find interesting things in one another'.

Reciprocal sentences with enclitic reflexive pronouns may be ambiguous with respect to number: те си пишат/te si pisat may mean 'they-two write to one another' or 'they-more-than-two write among themselves'. The ambiguity is eliminated with един друг/edin drug. Compare те намират интересни неща едни в други/te namirat interesni nesta

edni (PL) v drugi (PL) 'they (more than two) find interesting things in one another'.

In so far as they use the same pronouns, reflexive and reciprocal constructions are formally identical. The meaning of the verb often makes one or the other interpretation the more likely. Compare:

Мият се/Mijat se.

'(They) wash themselves.' (not '(They) wash each other.') Бият се./Bijat se.

'(They) hit one another.' (not '(They) hit themselves.')

Alternative readings are forced and ambiguities avoided with ceбe си/sebe si '-self' and един друг/edin drug 'one another':

Мият се един друг./Mijat se edin drug. (They-two) wash one another.' Себе си бият./Sebe si bijat. (They) beat themselves.'

4.9 Possession

Possession is expressed verbally with имам/imam '(I) have' and its negative counterpart нямам/пјатат '(I) don't have': Иван има много приятели/Ivan ima mnogo prijateli 'Ivan has many friends'. Possession in noun phrases is expressed in several ways:

- 1 Clitic dative personal pronouns (see section 3.1.3): това са книгите ми/tova sa knigite mi 'these are my books' (= 'this-N SG are books-the me-DAT').
- 2 Prepositional phrases with на/na: това е домът на брат ми/tova e domat na brat mi 'this is the home of my brother'.
- 3 Possessive personal pronouns (section 3.1.3): това са моите книги/ tova sa moite knigi 'these are my books'.
- 4 Possessive adjectives derived from personal nouns (section 3.3.2): това е братовият дом/tova e bratovijat dom 'this is (my) brother's home'. Adjectives of this sort are colloquial.

4.10 Quantification

The principal quantifiers of Bulgarian include cardinal numerals (see section 3.1.5), pronominal quantifiers (like колко/kolko 'how many, much') and много/mnogo 'many, much', малко/malko 'few, little', немного/nemnogo 'not many, much'.

Two cardinal numerals are inflected for gender: '1' is masculine, feminine or neuter, depending on the gender of its head; '2' contrasts masculine and feminine/neuter forms. (See section 3.1.5 for forms and examples.)

Used alone, '1' governs a singular noun: една книга/edna kniga 'one book'. In compounds '1' is regularly singular and the noun plural or, less commonly, singular: тридесет и една книги (книга)/trideset i edna (F SG) knigi (PL) (kniga (SG)) '31 books' (Граматика/Gramatika 2: 183). With all other quantifiers, feminine and neuter nouns are plural: много книги/mnogo knigi 'many books (F)', двадесет и две писма/dvadeset i dve pisma '22 letters (N)'. Masculine nouns with quantifiers make use of both normal and count plurals (see section 3.1.2):

- All masculine nouns use the normal plural with 'many', 'few' and 'not many' or with 'how many' in exclamations: много студенти/mnogo studenti 'many students', колко студенти!/kolko studenti! 'how 1 many students!'.
- Masculine personal nouns use the normal plural after cardinal numerals '2'-'6' in -(и)ма/-(i)ма: двама студенти/dvama studenti 'two students'; otherwise they use the ordinary plural or count plural (with a clear preference for the former): пет ученици/pet učenici ~ 2 пет ученика/pet učenika 'five pupils (PL ~ COUNT PL)'.

 Otherwise, masculine non-personal nouns use the count plural: три
- 3 града/tri grada 'three cities (COUNT PL)'.

'Incorrect' usage with masculine nouns is common; see Граматика/ Gramatika (2: 183) and Scatton (1984: 312–13).

Verbs with cardinal numerals as subjects are singular if the numeral is '1'

alone; otherwise they are plural:

Една жена дойде./Edna žena dojde. 'One woman came.' Две жени дойдоха./Dve ženi dojdoxa. 'Two women came.'

Personal pronouns are quantified with cardinal numerals. The pronoun occupies the first position; the numeral, generally definite, follows. Cardinals in -(и)ма/-(i)ma ('2'-'6') are used with groups of men or men and women: ние двамата ще дойдем/nie dvamata šte dojdem 'we two

and women: ние двамата ще доидем/піе dvamata ste dojdem 'we two (= two males or a male and a female) will come-1 PL'.

Quantification is also expressed by nouns denoting measurements (литър/lităr 'litre'), containers (чаша/са́за 'сир, glass'), collections of elements (стадо/stado 'herd'), parts (порция/рогсіја 'portion') and others. The quantifier precedes the noun, which is singular or plural as appropriate: литър мляко/lităr mljako 'a litre of milk', чифт волове/сіft volove 'a ратър охеп'. In these constructions verbal concord is usually with the quantifier: чашата вино е била на масата/čašata vino e bila na masata 'the glass-F SG of wine-N SG was-F 3 SG on the table'. However, groups show two types of concord: singular, focusing on the group as a

unit, or plural, focusing on the latter as a collection of individuals:

Накрая група байловчани излезе на сцената./Nakraja grupa bajlovčani izleze na scenata.

'Finally a group of inhabitants-of-Bajlovo came-3 sG on stage.'

Група младежи от Русе заминали за Съветския съюз./Grupa mladeži ot Ruse zaminali za Săvetskija sājuz.

'A group of young-people from Ruse left-3 PL for the Soviet Union.'

(Граматика/Gramatika 3: 151-2)

These quantifiers may themselves be quantified as ordinary nouns: две чаши вода/dve čaši voda 'two glasses of water'.

'Existential-be' is има/ima 'has'; its negative counterpart is няма/ njama 'hasn't'. They are invariably neuter third person singular and, as the syntax of pronominal forms shows, they take direct object complements:

Нямаше яйца./ Njamaše jajca.
'(There) weren't-3 sG eggs-PL.'
Има ги вкъщи./ Ima gi vkašti.
'They are at home.' (= 'has-3 sG them-ACC at-home')

5 Lexis

5.1 General composition of the word-stock

The standard 'Academy' dictionary of Bulgarian (Романски/Romanski 1955-9) contains over 63,000 entries, of which 25 per cent are foreign borrowings or words derived from them (Бояджиев/Војадžiev 1970). Of native lexical items, perhaps as many as 2,000 are directly inherited from Proto-Slavonic through Old and Middle Bulgarian (Русинов/Rusinov 1980: 76). Inherited words represent diverse lexical domains and include much of the most common, basic vocabulary of the language. The number of words derived from them is perhaps 15-20 times greater, in the order of 30-40,000.

The relative weight of inherited Proto-Slavonic material can be estimated from Николова/Nikolova (1987) – a study of a 100,000-word corpus of conversational Bulgarian. Of the 806 items occurring there more than ten times, approximately 50 per cent may be direct reflexes of Proto-Slavonic forms; nearly 30 per cent are later Bulgarian formations and 17 per cent are foreign borrowings or words derived from them. (The remaining items are problematical; figures mine, EAS.)

Russian is the only Slavonic language that has had any significant influence on Bulgarian. Large numbers of native Russian and Russian Church Slavonic words are common throughout the language (examples below). Until recently, Russian has been the principal intermediary source for much of Bulgarian's international vocabulary.

5.2 Patterns of borrowing

Of approximately 16,200 foreign words in Романски/Romanski (1955–9), a single original source can be found for about 14,500. Of these, 96 per cent are attributable to only eight languages, each accounting for at least 1 per cent or more:

1	Latin	25.5%	(3,700)
2	Greek	23.0%	(3,350)
3	French	15.0%	(2,150)
4	Turkish*		(1,900)
5	Russian	10.0%	(1,500)
6-7	Italian		(500)
6-7	German	3.5%	(500)
8	English	2.0%	(300)

^{*}including Arabic and Persian. Figures based on Бояджиев/Војадžіеч 1970.

The vast majority of Latin borrowings – mostly international terms in the areas of politics, civil administration, scholarship, law, medicine and others – entered the modern language through Russian or other, western languages: администрация/administracija 'administration', секретар/sekretar 'secretary'. (The material here is based on Русинов/Rusinov 1980: 76–94, and works cited there.) A few date from Old and Middle Bulgarian.

Greek lexical material first entered Bulgarian during the Old Bulgarian period, mostly in the areas of religion and civil and military administration: икона/ikona 'icon', деспот/despot. During the Ottoman period, strong Greek political and cultural influence facilitated borrowings not only of literary, but also common, popular lexical material: пирон/piron 'nail', евтин/evtin 'inexpensive'. During the Bulgarian Renaissance (eighteenth and nineteenth centuries), international vocabulary of Greek origin entered Bulgarian through Russian or other languages: граматика/gramatika 'grammar', демокрация/demokracija 'democracy'.

French borrowings appear from the beginning of the nineteenth century, first from Russian, later directly from French. They are concentrated in

first from Russian, later directly from French. They are concentrated in social and political life, military affairs, cooking, dress and the arts: aταιμέ/atašė 'attaché', διοφέτ/bjufėt 'buffet', τακτή/taksi 'taxi'.

Bulgarian borrowings from Turkish (and through Turkish, Persian and Arabic) were extremely numerous during the time of the Ottoman domination. One of the major features of the history of the modern Bulgarian literary language – particularly after 1878 – was the systematic replacement of Turkish borrowings with Bulgarian neologisms or borrowings from Russian or western European languages. Pomahckh/Romanski (1955–9) now lists only about 2,000 Turkish borrowings, of which about 800 (house-hold items occupations items of clothing foods plants animals) are styhold items, occupations, items of clothing, foods, plants, animals) are stylistically neutral: μοράπ/ἔοτὰρ 'sock', μάμτα/ἔἀπτα 'purse'. The remainder

are archaic, dialectal, non-standard and/or pejorative, and generally have neutral native synonyms: кютук/kjutuk (from Turkish) versus пън/pan '(tree) stump' (Лакова/Lakova 1972).

In so far as they do not take into account Russian Church Slavonic forms (often indistinguishable from native Bulgarian forms) or the intermediary role of Russian in the transmission of international terminology, the figures above substantially underestimate the impact of Russian on the Bulgarian lexicon. In early Modern Bulgarian, the influence of Russian was through Church Slavonic. By the end of the eighteenth and beginning of the nineteenth centuries, literary Russian was the predominant foreign the nineteenth centuries, literary Russian was the predominant foreign lexical influence on Bulgarian, and remained so until recently. Russian and Russian Church Slavonic lexical items are found in virtually every area of Bulgarian vocabulary: вселена/vselena 'universe', вероятен/verojaten 'probable', старая се/staraja se '(I) try', обаче/obace 'however'.

The source of several important formations is indeterminate: Old/

Middle Bulgarian literary sources and/or early printed Church books from Russia; for example, agent nouns in /-tel/ (учител/učitel 'teacher'), abstract nouns in /-ost/ (нежност/nežnost 'tenderness'), deverbative nouns in /-ie/ (предложение/predloženie 'proposal'), the present active participle (see section 3.2.1).

German borrowings (mostly military and technical) and Italian borrowings (concentrated in art, music, business and food) date from the nineteenth and twentieth centuries.

Prior to 1878 English borrowings were extremely limited; their numbers did not increase substantially until after the Second World War. Since then, and especially in the last twenty years, English has become the greatest foreign lexical influence on Bulgarian, and this influence is now far greater than the above figures (based on work more than thirty years old) suggest. The impact of English is particularly strong in technology, tourism, sports, dress, the arts and music and popular culture: транзистор/tranzistor 'transistor', мотел/motel, джинси/džinsi 'jeans', танк/tank.

Incorporation of borrowings 5.3

Borrowings are generally adapted to the phonological and morphological systems of Bulgarian. Phonological adaptation is limited to eliminating segments not found in the Bulgarian phonemic system or to correcting violations of sequential constraints. For example, geminate consonants are regularly replaced by single phonemes: Russian грамматика/grammatika becomes Bulgarian граматика/gramatika 'grammar'. Front round vowels are replaced by sequences of palatalized or palatal consonants plus back vowels: French bureau gives бюро/bjuro 'office'. Russian palatalized consonants are eliminated where they do not occur in Bulgarian: Russian пятилетка/pjatiletka /p'it'il'etka/ gives Bulgarian петилетка/petiletka /petiletka/ 'five-year plan'.

The last example illustrates an accommodation characteristic of borrowings from Russian and Russian Church Slavonic: the replacement of Russian morphological components with their Bulgarian counterparts, in this case Bulgarian πet/pet for Russian πάτь/pját 'five'.

While showing some anomalies (below), borrowings are generally adapted morphologically. Nouns are assigned to one of the major gender classes, generally on the basis of their final segments. Nouns terminating in consonants are masculine (English ketchup – κέττυπ/κέτčάρ); those terminating in /a/ are feminine (Rumanian masã – maca/masa 'table'). Nouns terminating in /o e u i/ are neuter if non-personal (French bureau – διορό/bjurò 'office') or masculine if personal (French attaché – ataшe/ atase 'attaché').

Other, more complex types of adaptation also occur. For example, French feminine nouns ending in 'mute e' are regularly borrowed as feminine nouns in /-a/: allée - anen/aleja 'avenue, lane'. Greek and Latin borrowings show complex suffix alterations and shifts among inflectional classes (Първев/Părvev 1979).

The inflectional behaviour of borrowed nouns may be anomalous. A fairly large number of the monosyllabic masculine nouns that take the plural suffix /-i/ (instead of /-ove/) are foreign (see section 3.1.2). Similarly, the velar-dental alternation accompanying the masculine plural marker /-i/ is often absent (sections 2.2, 3.1.2).

Borrowed adjectives are altered in order to agree with the shape of **derived** native adjectives. This entails little change in already derived Russian adjectives: Russian кольхозный/kol'xoznyj – Bulgarian колхозен/kolxozen 'collective-farm (M SG)'. Otherwise, a Bulgarian suffix is commonly added to the foreign adjective: English *loyal* – лоялен/lojalen.

Borrowed adjectives are generally inflected like native adjectives. However, in the past, many adjectives borrowed from Turkish without the addition of a derivational suffix had a single unchanging form; the few that remain continue to do so (see section 3.1.4). Unmodified, uninflected adjectives from other sources are still possible, particularly in the colloquial language. Many are ephemeral, and even those that survive show syntactic idiosyncrasies (Πървев/Părvev 1979: 233f.). For example, rporu/grogi (from English groggy) is used only predicatively: той е гроги/toj e grogi 'he is groggy'.

Borrowed verbs are most commonly assimilated into the third conjugation with the suffix /-ira-/ (of German origin): English *train* – тренирам/treniram '(I) train'. Most verbs of this type are bi-aspectual (see section 3.2.1). There is a tendency in colloquial Bulgarian to integrate them into the aspectual system by creating aspectual mates for them with perfectivizing prefixes or the imperfectivizing suffix /-va-/ (Граматика/ Gramatika 2: 268).

The third-conjugation suffix /-uva-/ is used to assimilate Russian verbs in /-ovat'/: арестовать/arestovat' - арестувам/arestuvam '(I) arrest (IMPFV and PRFV)'. Many of these verbs are bi-aspectual.

5.4 Lexical fields

5.4.1 Colour terms

Basic colour terms are given in bold; examples are masculine singular.

'white' **бя́л/bjal** 'black' **че́рен/čéren**

'red' червен/červen; also ален/alen 'scarlet'

'green' **зеле́н/zelén**

'yellow' жълт/žalt; also pyc/rus 'blond (of hair)'

'blue' cин/sin; also лазурен/lazuren 'azure, sky-blue (poetic)'

and електрик/elektrik 'electric blue (indeclinable)'

'brown' кафяв/kafjav (< кафе/kafe 'coffee'); also кестеняв/

kestenjav 'auburn, chestnut (of eyes, hair, horses)' («кестен/kesten 'chestnut'), бежов/bezov 'beige'

'purple' морав/morav; also виолетов/violetov 'violet',

пурпурен/purpuren 'crimson (literary, poetic)'

'pink' poзов/rozov (v poза/roza 'rose'); also пембян/pembjan

~ пембен/pemben 'hot pink'

'orange' opанжев/oranžev

'grey' cив/siv

5.4.2 Body parts

'head' глава́/glavá

'eye' око/oko (anomalous plural: 3.1.2)

'nose' нос/nos

'ear' yxo/uxo (anomalous plural: 3.1.2)
'mouth' ycτά/usta; note related 'lip' ycτηα/ustna

'hair' косъм/kosam 'single hair'; коса/kosa 'head of hair'

'neck' врат/vrat, шия/šija

'arm, hand' ръка/raka (anomalous plural: 3.1.2); also длан/dlan

'palm'

'finger' пръст/prast

'leg, foot' κράκ/krák (anomalous plural κρακά/kraká: 3.1.2); also

'(sole of) foot' стъпало/stăpalo, ходило/xodilo

'toe' пръст/prast (with на крака/na kraka 'of the foot' to

distinguish 'toe' from 'finger')

'chest' гърди/gardi (plural of гръд/grad 'breast' (F)') and

гръден кош/gråden koš (literally 'breast basket')

'heart' сърце́/sărcé

5.4.3 Kinship terms

'mother' ма́йка/majka, and for one's own mother ма́ма/máma 'father' баща/bašta, and for one's own father та́тко/tátko

'sister' сестра/sestra; also кака/kaka 'older sister'

'brother' брат/brat (anomalous plural братя/bratja); also батко/

batko 'older brother'

'aunt' ле่ля/lelja 'sister of father or mother', вуйна/vujna 'wife of mother's brother', стрина/strina 'wife of father's

brother'; also dialectal TeTKa/tetka 'sister of mother'

'uncle' чичо/čičo 'brother of father' and вуйчо/vujčo 'brother

of mother'; also dialectal свако/svako, тетин/tetin, лелин/lelin 'husband of mother's sister'. Леля/lelja 'aunt' and чичо/čičo 'uncle' are used when the more precise relationship is unknown and also to address adults

in general.

'niece' племенница/plemennica; also dialectal сестриница/

sestrinica 'daughter of sister' and братаница/bratanica

'daughter of brother'

'nephew' племенник/plemennik; also dialectal сестриник/

sestrinik 'son of sister' and братанец/bratanec 'son of

brother'

'cousin' female братовчедка/bratovčedka and male

брадовчед/bratovčed

'grandmother' ба́ба/baba; also used to address elderly women 'grandfather' дя́до/djado; also used to address elderly men

'wife' жена/žena (also 'woman'), съпруга/sapruga ('spouse-F') 'husband' мъж/maž (also 'man'; anomalous plural: 3.1.2),

съпруг/săprug ('spouse-м')

'daughter' дъщеря/dăšterjà (anomalous vocative: 3.1.2)

'son' cин/sin

6 Dialects

Bulgarian dialects are usually divided into west and east with respect to the development of late Proto-Slavonic *ě jat'. West of a line running north—south between A and B on map 5.1, *ě became /e/: 'big' M SG /golėm/, PL /golėmi/, 'milk' /mlėko/, 'milkman' /mlekar/. East of the line, stressed *ě in some (if not all) environments either becomes /a/ (as in the literary language; see section 2.3) or retains what may have been one of its late Proto-Slavonic vocalizations, /a/; unstressed /e/ (reduced phonetically to [i], see below) is general: /gol'am gol'em'i ml'ako ml'ekar/ (Стойков/ Stojkov 1968: 54–5 for details).

While not coinciding exactly with the *jat'* boundary, other important isoglosses generally reinforce the east-west division:

- 1 Late Proto-Slavonic *a after alveo-palatal consonants (including j) gives *ě in the east, but /a/ in the west (as in the literary language): 'frog' east F SG /žaba/ PL /žeb'i/ versus west /žaba/ /žabi/.
- 2 Strong reduction of unstressed /a/ to [a], /o/ to [u], /e/ to [l] is typical in the east, especially north-east; in the west it is lacking altogether or limited to /a/ > [a] (as in the literary language; see section 2.1).
- 3 Late Proto-Slavonic * ρ and * δ give δ in the east (as in the literary language; section 2.1); in the west both δ and δ are common.
- East Bulgarian commonly has palatalized labial and dental consonants before front vowels and at the ends of words (where they contrast with non-palatalized consonants): /z'et'/ 'son-in-law'. In the west only /l' n' t' d'/ are common, with /l' t' d'/ often replaced by /j' k' g'/; otherwise consonants are non-palatalized (as in the literary language; section 2.1).
- 5 Proto-Slavonic *tj and *dj are /št/ and /žd/ in the east (the literary treatment; section 2.1); besides these also / \check{c} $\check{3}$ /, /k' g'/, and / $\check{s}\check{c}$ $\check{z}\check{3}$ / in the west.
- 6 Word-internal alternations of /ră ~ ăr/, /lă ~ ăl/ (see section 2.3) are characteristic of eastern dialects; in the west one finds syllabic liquids, non-alternating sequences of vowel-liquid or liquid-vowel, or in the case of /l/, replacement by /ă/ or /u/.
- 7 In the east present-tense verb forms are as in the literary language (see section 3.2). In the west, palatalization is often lost in the first person singular and third-person plural: /xoda/ '(I) go'. Alternatively, the first person singular may be formed with /-m/: /xodim/. In the west /-me/ commonly marks the first person plural in all conjugations: /xodime/ '(we) go'.
- 8 The imperfect past active participle (see section 3.2.1) is typically eastern. Clearly, a number of these features bring western dialects closer to Serbo-Croat and Macedonian.

The most archaic Bulgarian dialects are found in south-eastern Bulgaria, in the Rhodope mountains. Important features (Стойков/Stojkov 1968: 87-91) include:

- 1 /ä/ as the reflex of *ě and of *a after alveo-palatals: /gol'äm/, /gol'äm'i/, /žäba/;
- 2 a single reflex /a/, /a/, /o/ or /o/ for stressed PSl. *ρ, τω, ε, ω, with palatalized labials and dentals before the reflexes of *ε and ω: *ρ /raka/ 'hand, arm', *τω /daš/ 'rain', *ε /m'aso/ 'meat', *ω /l'asno/ 'easily';
- 3 'triple' definite article: unmarked /-t-/, /-s-/ for proximity to the speaker, /-n-/ for distance from the speaker (compare Macedonian);
- 4 numerous case vestiges in nouns, adjectives and definite articles.

Bulgarian dialects Map 5.1 Danube Romania Danube Varna Serbia Tărnovo Balkan Range Black Sea Sofia Burgas Pazardjik Maritsa Plovdiv Rhodopes Pirin Macedonia Turkey • Smoljan A-B: West vs. East Greece

Source: Стойков/Stojkov 1968: 63

This dialect picture is largely **historical**, reflecting the speech of informants at least 50-60 years of age when the data of the major dialectological works of the past 30-40 years were collected (Стойков/Stojkov 1964-81). With the passing of this generation, the spread of literacy, the growth of the mass media and the rapid urbanization of the last twenty years, the older picture has given way to a much different one mixing traditional features and the standard language.

The geographical extent of Bulgarian dialects is controversial. On the level of local dialects there is no sharp boundary between the speech of western Bulgaria and that of eastern Serbia, former Yugoslav Macedonia and areas of Greece and Turkey contiguous to Bulgaria in which Slavonic dialects are still spoken. The official Bulgarian position, with respect to dialects and earlier historical periods, has been that eastern Serbian dialects, all Macedonian dialects in former Yugoslavia and Slavonic dialects in Turkey and Greece are dialects of Bulgarian.

Acknowledgement

I wish to acknowledge the many suggestions, stylistic and substantive, of Borjana Velčeva and Linda Scatton. Special thanks are due to Vladimir Filipov, my Bulgarian-language consultant, whose assistance went far beyond that of a native informant.

References

Aronson, Howard I. (1967) 'The grammatical categories of the indicative in the contemporary Bulgarian literary language', in *To Honor Roman Jakobson*, vol. I, The Hague: Mouton, 82–98.

—— (1968) Bulgarian Inflectional Morphophonology, The Hague: Mouton.

Chvany, Catherine V. (1988) 'Distance, deixis and discreteness in Bulgarian and English verb morphology', in Alexander M. Schenker (ed.) American Contributions to the Tenth International Congress of Slavists, Sofia, September 1988, Linguistics, Columbus, Oh.: Slavica, 69-90.

Ewen, Robert C. (1979) 'A grammar of Bulgarian clitics', Ph.D. dissertation, Seattle: University of Washington. (Available from Ann Arbor: University

Microfilms International.)

Hauge, Kjetil Rå (1976) The Word Order of Predicate Clitics in Bulgarian (Meddelelser, 10), Oslo: Slavic-Baltic Institute, Oslo University.

Lindstedt, Jouko (1985) On the Semantics of Tense and Aspect in Bulgarian, Helsinki: University of Helsinki.

Mayer, Gerald L. (1988) The Definite Article in Contemporary Standard Bulgarian, Wiesbaden: Harrassowitz.

Rudin, Catherine (1986) Aspects of Bulgarian Syntax: Complementizers and WH Constructions, Columbus, Oh.: Slavica.

Scatton, Ernest A. (1975) Bulgarian Phonology, Cambridge, Ma.: Slavica. (Reprinted. Columbus, Oh.: Slavica, 1983.)

(1984) A Reference Grammar of Modern Bulgarian, Columbus, Oh.: Slavica.

- Velcheva, Boryana (1988) Proto-Slavic and Old Bulgarian Sound Changes, Columbus, Oh.: Slavica.
- Бояджиев, Симеон (1970) 'За лексиката от чужд произход в книжовния български език', Известия на Института за български език 19: 403-7.
- Дончева, Лиляна (1975) 'Наблюдения върху квантитативния аспект на номинативните лични местоимения в руски и в български език', Бъглгарски език 25, кн. 1: 9–20.
- Георгиева, Ел. и Станков, В., отговорни редактори (1983) Правописен речник на съвременния български книжовен език, София: Българска академия на науките.
- Граматика на съвременния български книжовен език (1982-3) Т. І: Фонетика, Димитър Тилков, редактор, Т. II: Морфология, Стоян Стоянов, редактор, Т. III: Синтаксис, Константин Попов, редактор, София: Българска академия на науките.
- Лакова, Мери (1972) 'Турцизмите в "Речник на съвременния български книжовен език" от стилистична гледна точка', *Български език* 22, кн. 1–2: 61–72.
- ——— (1978) 'Въпросно-отговорни единства в съвременния български книжовен синтаксис', *Български език* 28, кн. 5: 418–34.
- Маслов, Ю. С. (1981) Грамматика болгарского языка, Москва: Высшая школа.
- Николова, Цветанка (1987) Честотен речник на българската разговорна реч, София: Наука и изкуство.
- Пашов, Петър (1966) Българският глагол, София: Наука и изкуство.
- Пашов, Петър и Първев, Христо (1979) Правоговорен речник на българския език, София: Наука и изкуство.
- Пенчев, Йордан (1984) Строеж на българското изречение, София: Наука и изкуство.
- Първев, Христо (1979) Помагало по българска лексикология, София: Наука и изкуство.
- Романски, Стоян, главен редактор (1955-9) *Речник на съвременния български книжовен език*, 3 тома, София: Българска академия на науките.
- Русинов, Русин (1980) Речниковото богатство на българския книжовен език, София: Народна просвета.
- Стойков, Стойко, редактор (1964-81) Български диалектен атлас, 4 тома, София: Българска академия на науките.
- —— (1968) Българска диалектология, София: Наука и изкуство.
- Тилков, Димитър и Бояджиев, Тодор (1981) Българска фонетика, София: Наука и изкуство.
- Чолакова, Кристалина, главен редактор (1977–90) Речник на българския език, 6 тома [а й], София: Българска академия на науките. In progress.